The asymmetry in firms' mechanisms of cash holdings adjustments: evidence from the G-5 economies

Cuong Nguyen

Review of Quantitative Finance and Accounting

ISSN 0924-865X

Rev Quant Finan Acc DOI 10.1007/s11156-018-0754-1 Review of Quant Finance and Account

EDITOR:

Cheng-few Lee ASSOCIATE EDITORS:

Richard T. Baillie Randolph P. Beatty Ivan Brick Lawrence D. Brown Stephen Brown Vihang Errunza Wayne E. Ferson Joseph E. Finnerty Lawrence R. Glosten Ravi Jagannathan Bikki Jaggi Kose John Richard E. Kihlstrom E. Han Kim Kenneth Lawrence Haim Levy Wilbur G. Lewellen Andrew W. Lo Timothy J. Loughran Robert McDonald James C. McKeown

Larry J. Merville Paul Newbold James Ohlson Man-Young Park Stephen H. Penman Joshua Ronen Ehud Ronn Michael S. Rozeff Robert A. Schwartz Lemma W. Senbet Jay Shanken David B. Smith Richard J. Sweeney Sheridan Titman Charles Tzcinka Robert E. Versecchia an Chunchi Wu Paul A. Zarowin

Springer VOLUME 24, NUMBER 1 FEBRUARY, 2005





Your article is published under the Creative Commons Attribution license which allows users to read, copy, distribute and make derivative works, as long as the author of the original work is cited. You may selfarchive this article on your own website, an institutional repository or funder's repository and make it publicly available immediately.



ORIGINAL RESEARCH



The asymmetry in firms' mechanisms of cash holdings adjustments: evidence from the G-5 economies

Cuong Nguyen¹

© The Author(s) 2018

Abstract

Using a G-5 country sample (France, Germany, Japan, the UK, and the US) from 1980 to 2007, I find new evidence of the asymmetry in firms' mechanisms of cash holdings adjustments. They undertake different approaches to move toward their target cash holdings levels conditional on whether they have below- or above-target cash holdings. Specifically, firms with above-target cash holdings adjust mainly via changes in cash flows from financing and investing. They generally reduce their levels of equity proceeds, net debt issues and fixed asset disposal but increase their levels of equity repurchases, dividend payout, net assets from acquisitions, portfolio and short-term investments, and capital expenditures. However, firms with below-target cash holdings adjust mainly via changes in operating cash flows, i.e., they increase their levels of funds from operations but reduce their levels of working capital. The mechanisms undertaken by firms with above-target cash holdings allow them to adjust toward their target cash holdings relatively faster than those with below-target cash holdings, as they possibly incur lower costs than the mechanisms experienced by firms with below-target cash holdings. The results highlight the importance of understanding the asymmetry in firms' mechanisms of cash holdings adjustments when analyzing how they adjust toward their target cash holdings levels.

Keywords Cash holdings \cdot Adjustment mechanisms \cdot Adjustment costs \cdot Asymmetric adjustments

JEL Classification $G30 \cdot G31 \cdot G32$

Cuong Nguyen M.Nguyen@hud.ac.uk

¹ Huddersfield Business School, The University of Huddersfield, Queensgate, Huddersfield HD13DN, UK

1 Introduction

In the corporate cash holdings literature, firms' cash holdings behaviors have been examined using frameworks largely similar to those employed by studies on corporate capital structures. For example, following the trade-off theory of corporate capital structures, the trade-off view¹ holds that firms derive their target levels of cash holdings through balancing between the marginal benefits of cash (e.g., the avoidance of costs of transaction, adverse selection, and agency of external financing and a better ability to undertake investment opportunities) and its marginal costs (e.g., opportunity costs of holding low-return assets and the increase in marginal tax rates and agency costs of managerial discretion) and actively adjust toward these over time² (Opler et al. 1999; Ozkan and Ozkan 2004; Dittmar and Duchin 2010; Venkiteshwaran 2011; Jiang and Lie 2016; Orlova and Rao 2018). In addition, their adjustment speeds are generally swifter when firms have above-target cash holdings i.e., their actual levels are higher than their target levels of cash holdings than when they have below-target cash holdings i.e., their actual levels are lower than their target levels of cash holdings possibly due to lower adjustment costs (Dittmar and Duchin 2010; Jiang and Lie 2016; Orlova and Rao 2018). However, very little has been known about the mechanisms undertaken by firms to move toward their target levels of cash holdings. This paper aims to provide a comprehensive examination of the mechanisms firms choose to undertake to move toward their target levels of cash holdings conditional on whether they have below- or above-target cash holdings using a G-5 country sample (France, Germany, Japan, the UK, and the US). Yet, it seems intuitive that different firms do not undertake similar mechanisms to move toward their target levels of cash holdings as they are subject to different levels of costs of deviations from target cash holdings and costs of adjustments. Examining such mechanisms may provide a sound justification for the asymmetry in firms' speeds of adjustments reported in the existing corporate cash holdings literature.

Although there has been some research on firms' asymmetric adjustments toward target cash holdings i.e., different speeds of adjustments contingent on the position relative to their target cash holdings which are driven by differences in costs of deviations from target cash holdings which are defined as the differences between firms' target levels and their actual levels of cash holdings, costs of adjustments, and levels of financial constraints (Opler et al. 1999; Dittmar and Duchin 2010; Venkiteshwaran 2011; Jiang and Lie 2016; Orlova and Rao 2018), very little has been known about the mechanisms they may undertake to adjust toward their target cash holdings. Dittmar and Duchin (2010), Jiang and Lie (2016) and Orlova and Rao (2018) show that firms with excess cash or above-target cash holdings may adjust toward their target cash holdings faster than those with a cash shortage due to lower costs of adjustments. In contrast, (Venkiteshwaran 2011) finds that

¹ Different from the trade-off view, the financing hierarchy view which follows the pecking-order theory of corporate capital structures suggests that firms have no optimal levels of cash holdings (Myers and Majluf 1984). There is a negative relation between their cash balances and investment opportunities as firms prefer to finance their investment opportunities with internal funds for information asymmetries may make external financing become expensive.

 $^{^2}$ In line with studies on leverage rebalancing (Leary and Roberts 2005; Strebulaev 2007) that imperfect capital markets or costs of adjustments may prevent firms from undertaking continuous adjustments to rebalance their leverage, Ozkan and Ozkan (2004) find that on average, UK firms may adjust toward their target cash holdings levels at speeds of from 54 to 60% each year which are comparable to those reported by Venkiteshwaran (2011) (higher than 50%) but faster than those reported by Opler et al. (1999) (from 33 to 35%) and Dittmar and Duchin (2010) (from 20 to 40%) for US firms.

cash-deficient firms or those with below-target cash holdings may adjust more quickly due to higher costs of deviations caused by their financial constraints. These studies consider the role of financial constraints, costs of adjustments and costs of deviations from targets on firms' speeds of adjustments which are more or less related to adjustments in cash flows from financing (CFF) i.e., adjustments are made through debt issue/retirement and/or equity issue/repurchase decisions. The role of costs associated with adjustments in cash flows from operating (CFO) and cash flows from investing (CFI), however, has received almost no attention. This is surprising since firms may adjust toward their target cash holdings through adjustments in not only CFF but also CFO and CFI. For example, in addition to distributing some of their excess cash to shareholders through dividend increases or stock repurchases and retiring part of their debt, firms with excess cash may increase their net working capital (e.g., stocking more inventories, piling up more accounts receivable and reducing accounts payable) and investments (e.g., increasing capital expenditures and acquisitions)—mechanisms opposite to those undertaken by cash-deficient firms. Previous studies tend to treat these adjustments as exogenous, independent of firms' cash and leverage management policies. I now explicitly account for these. I investigate why firms' speeds of adjustments may vary contingent on the position relative to their target cash holdings by examining the possible mechanisms they may undertake to adjust toward their target cash holdings which include adjustments in all of the three cash flow categories.

There may be certain costs associated with adjustments in *CFO*. A decrease in firms' working capital, for instance, may involve an increase in their accounts payable and/or a reduction in their accounts receivable and inventories. Increasing accounts payable may lead to worsened credit reputation, forgone cash discounts, increased administration costs and prices set by suppliers, late payment penalties, and damaged relationships with suppliers (Wu et al. 2012). According to García-Teruel and Martínez-Solano (2010), supplier financing may turn out to be more costly than other sources of financing and firms therefore try to avoid using it when possible. Meanwhile, a reduction in accounts receivable may imply stricter credit terms which then result in declined sales and worsened relationships with customers (Brealey et al. 2014). Finally, when inventories are reduced not because of better inventory management, firms may be unable to fill up their customers' orders, suggesting high inventory shortage costs.

Similarly, adjustments in *CFI* may incur significant costs. Managers of cash-rich firms are likely to make value-decreasing acquisitions (Harford 2002; Harford et al. 2012) as their compensation is usually increased after these (Bliss and Rosen 2001; Harford and Li 2007). Costs of adjustments hence may be embodied in a decrease in the value of these firms. When cash-deficient firms have to give up positive *NPV* (net present value) investments, these costs may be the opportunity costs arising from forgone future cash flows generated by these investments, as implied by the pecking-order view.

Examining relevant costs associated with firms' cash holdings adjustments in the G-5 countries (France, Germany, Japan, the UK, and the US) during the 1980–2007 period, consistent with the existing literature, I find that there is asymmetry in firms' adjustments. Firms with above-target cash holdings move toward their target cash holdings faster than those with below-target cash holdings. In addition, the most important finding of my study is that there is asymmetry in the mechanisms firms undertake to move toward their target cash holdings. Specifically, I find new evidence that in addition to adjustments in *CFF*, firms also adjust toward their target cash holdings through adjustments in *CFO* and *CFI*. Firms with excess cash generally reduce *CFF* and *CFO* and increase *CFI*—mechanisms opposite to those undertaken by firms with below-target cash holdings may incur lower costs for three

reasons. First, costs associated with debt retirements, equity repurchases, and dividend increases (*CFF*) may be lower than those associated with debt and equity issues and dividend reductions (Leftwich and Zmijewski 1994; Byoun 2008). Second, a decrease in *CFO* is less likely to lead to costs associated with operational problems. Third, decreasing *CFI* or having to give up investment opportunities (e.g., to decrease capital expenditures and acquisitions) may incur huge opportunity costs associated with permanently forgone future cash flows generated by these. These together explain why firms with above-target cash holdings may adjust toward their targets faster.

I perform several additional analyses and find evidence on the influence of the magnitude of firms' deviations from targets on their target adjustments. In addition to the variation in costs of adjustments caused by different mechanisms of adjustments, I argue that the magnitude of firms' deviations from target cash holdings is also likely to asymmetrically influence their adjustments for two reasons. First, the fixed cost component (e.g., fixed costs of debt/equity issues, permanent losses of customers and growth opportunities, etc.) associated with cash holdings adjustments may suggest that firms with too much excess cash or a large cash shortage can adjust toward their targets faster due to lower incremental costs when that fixed component can be effectively "shared" with transaction costs that incur when firms offset their large cash holdings imbalances. In contrast, those that are not sufficiently far away from their target cash holdings may have fewer incentives to offset their cash holdings imbalances. Second, agency theories would suggest that firms with too much excess cash may face higher costs associated with the free cash flow problem and hence should disgorge some of their cash, thus implying faster speeds of adjustments. Those with a large cash shortage may also adjust toward their target cash holdings faster for larger costs of deviations from target cash holdings may be in the forms of operational problems and forgone positive NPV investments. Consistent with the "shared" fixed cost argument, I find that firms with large deviations from targets adjust toward their targets faster as they may face lower incremental costs of adjustments. In addition, in support of agency theories, compared to firms with small excess cash holdings, firms with large excess cash holdings experience faster speeds of adjustments as the free cash flow problem may be more significant for them. Firms with a large cash shortage also adjust more quickly than those with a small cash shortage for they are more likely to suffer from operational problems and have to give up positive NPV investments.

In another analysis, I examine how firms' cash holdings adjustments may be asymmetrically influenced by their major characteristics such as their precautionary motives (proxied by firm size, cash flow volatility and dividend payout), financial constraints (proxied by growth opportunities and dividend payout), and corporate life-cycle (proxied by firm age). There is evidence that firms experience faster speeds of adjustments in the presence of stronger precautionary motives (smaller firm size, higher cash flow volatility and lower dividend payout), greater financial constraints (more growth opportunities), and lower levels of maturity (younger age). However, when these factors are interacted with deviations from targets, their impact becomes relatively less significant (except for firm age), suggesting firms' deviations from targets may be the major driver of their target adjustments.

Overall, I contribute to the corporate cash holdings literature by studying the asymmetry in the mechanisms firms undertake to move toward their target cash holdings which helps to provide a better understanding of why speeds of adjustments may vary contingent on whether firms have below- or above-target cash holdings. Contingent on having either below- or above-target cash holdings, firms may undertake different adjustments in the three groups of cash flows which may involve different levels of costs, thus leading to different speeds of adjustments i.e., firms with above-target cash holdings adjust toward their targets faster than those with below-target cash holdings as their mechanisms of adjustments may involve lower costs. The evidence on the asymmetry in the adjustments in *CFO* and *CFI* undertaken by these two groups of firms is especially novel. To the best of my knowledge, my study is the first attempt in the literature to look at that asymmetry.

The remaining of the paper is organized as follows. Section 2 presents the empirical models. Section 3 discusses the sample. Section 4 reports the main findings. Section 5 reports the additional empirical analyses. Section 6 concludes.

2 Empirical models and methods

To examine the asymmetry in the mechanisms undertaken by firms to move toward their target cash holdings conditional on the position relative to their target cash holdings, I first classify the sample firms into two groups with one consisting of firms with below- and another consisting of firms with above-target cash holdings. To do that, firms' target levels of cash holdings (CH_{it}^*) must be estimated and then compared to their actual levels of cash holdings to see if they have below- or above-target cash holdings (Dittmar and Duchin 2010; Jiang and Lie 2016; Orlova and Rao 2018). The unobserved CH_{it}^* is a function of firms' major fundamentals, as follows:

$$CH_{it}^* = \hat{\beta}' \mathbf{x}_{it},\tag{1}$$

where $\hat{\beta}'$ is a vector of parameters estimated from a fixed-effects regression of cash holdings on a vector of relevant determinants, \mathbf{x}_{ii} :

$$CH_{it} = \beta \mathbf{x}_{it} + \varepsilon_{it}.$$
 (2)

As guided by previous studies in the current literature (Opler et al. 1999; Ozkan and Ozkan 2004; Bates et al. 2009; Jiang and Lie 2016; Orlova and Rao 2018), I include in \mathbf{x}_{it} eight independent variables consisting of growth opportunities, firm size, cash flow volatility, cash flow, capital expenditures, net working capital, research and development expenses, and a dividend dummy to capture whether firms pay dividends or not. ε_{it} is an error component that includes firms' fixed effects and time effects and an *i.i.d.* error term. The firms' fixed effects control for time-invariant unobservable, unique firm and/or industry characteristics that cannot be captured by \mathbf{x}_{it} while the time effects allow us to examine the extent at which the change in firms' cash holdings can be effectively explained by changes in their fundamentals rather than the evolution of time (Bates et al. 2009). The relations between these eight variables included in \mathbf{x}_{it} and firms' target cash holdings are as follows.

Growth opportunities (GO) This is the sum of the market value of equity or market capitalization and the book value of total debt scaled by the book value of total assets. The precautionary motive argument suggests a positive relation between firms' growth opportunities and their cash holdings since costs associated with adverse cash flow shocks and financial distress may be more significant among high-growth firms (Opler et al. 1999; Bates et al. 2009). Due to more information asymmetries, high-growth firms are more likely to give up positive *NPV* investments in the presence of a cash shortage and costly external financing, thus needing to hold more cash. In contrast, low-growth firms should hold less cash to reduce the free cash flow problem (Jensen 1986).

Firm size (FS) This is the natural log of the book value of total assets measured in 1980 US\$ value. There may be a negative relation between firms' size and their cash holdings for three reasons. First, the transaction cost motive argument suggests that large firms

should hold less cash due to the presence of economies of scale i.e., lower transaction costs when their noncash financial assets are converted into cash (Baumol 1952; Miller and Orr 1966; Mulligan 1997). Second, as large firm size indicates less financial distress, fewer information asymmetries and hence better access to external capital markets, in the spirit of the precautionary motive argument (Almeida et al. 2005), large firms can hold less cash. Finally, Sufi (2009) shows that large firms may have more access to bank lines of credit—a close alternative to cash, thus having less need to hoard it.

Cash flow volatility (CFV) This is the absolute value of the difference between the first difference of cash flow (% change) and the average of first differences. Consistent with the precautionary motive argument, firms with riskier cash flows should accumulate more cash to cope with potential adverse cash flow shocks in the presence of costly external financing better (Opler et al. 1999; Bates et al. 2009). This indicates a positive relation between this variable and firms' cash holdings.

Cash flows (CF) This is operating income before depreciation minus interest expense, income taxes and total dividends paid scaled by the book value of total assets. Firms with stronger cash flows may have a better ability to accumulate more cash, as suggested by the pecking-order view. In addition, these firms may be also those with more growth opportunities and hence need to hold more cash (Opler et al. 1999; Bates et al. 2009). These together suggest a positive relation between cash flows and cash holdings.

Capital expenditures (CAPEX) This is capital expenditures scaled by the book value of total assets. There are two opposite predictions on the impact of this variable on firms' target cash holdings. On the one hand, in line with the pecking-order view, Riddick and Whited (2009) find that firms which are making significant investments in assets may experience a temporary fall in their cash. In addition, when capital expenditures lead to an increase in assets that can serve as collaterals better (e.g., fixed assets) and hence firms' debt capacities, they may have less need to hoard cash. These together suggest that cash holdings may be inversely related to capital expenditures. On the other hand, according to the trade-off view, it is possible that firms which are making significant capital expenditures may be high-growth ones, thus needing to hold more cash.

Net working capital (NWC) This is the difference between working capital and cash and cash equivalents scaled by the book value of total assets. As net working capital includes highly liquid assets that can be considered as close substitutes for cash (e.g., inventories and accounts receivable), firms with a higher level of net working capital are likely to hold less cash (Bates et al. 2009).

Research and development expenses (R&D) This is research and development expenses scaled by total sales. Since these expenses may be considered as a close proxy for growth opportunities and hence costs associated with financial distress and adverse cash flow shocks, the trade-off view suggests firms which are incurring more research and development expenses should hold more cash. Consistent with this view, Brown and Petersen (2011) show that in the presence of financing frictions, firms tend to depend intensively on cash to smooth their research and development expenses for adjustments in these expenses usually involve high costs (e.g., wages of highly skilled technology workers). In contrast, the financing hierarchy suggests a negative relation between such expenses and cash holdings as firms which are incurring large amounts of these may temporarily experience a fall in their cash balances.

Dividend dummies This variable is defined to be 1 for firms that pay dividends and 0 otherwise. Fazzari et al. (1988) find that financially unconstrained (possibly cash-rich) firms are more likely to pay dividends than financially constrained firms. Almeida et al. (2005) and Bates et al. (2009), however, show that firms that pay dividends may hold less

cash for they may be considered by investors as less risky and hence have better access to external capital markets.

After estimating firms' target cash holdings, I define firms as having above- or belowtarget cash holdings if their cash ratios in the last accounting period (CH_{it-1}) are higher than or equal to or lower than their target cash holdings levels in the last accounting period (CH_{it-1}^*) i.e., $CH_{it-1} \ge CH_{it-1}^*$ or $CH_{it-1} < CH_{it-1}^*$. The prior literature shows that firms' speeds of adjustments vary conditional on whether they have below- or above-target cash holdings. To see how deviations from target cash holdings may asymmetrically influence firms' target cash holdings adjustments, I follow the prior literature by introducing the following asymmetric partial adjustment model to account for the variation in firms' speeds of adjustments.

$$\Delta CH_{it} = \lambda_0 + \lambda_1 CDev_{it}CH^a_{it} + \lambda_2 CDev_{it}CH^b_{it} + \varepsilon_{it}, \tag{3}$$

where ΔCH_{it} is the difference between firms' cash ratios (e.g., cash and cash equivalents scaled by total assets) for the current (CH_{it}) and the last (CH_{it-1}) accounting periods. $CDev_{it}$ is deviations from target cash holdings and is defined as the difference between firms' target cash holdings, CH_{it}^* and CH_{it-1} ($CDev_{it} = CH_{it}^* - CH_{it-1}$). CH_{it}^a (CH_{it}^b) indicates that firms have above-target (below-target) cash holdings in the last accounting period. Within 1 year, firms partially adjust toward their target cash holdings at the speed of adjustment λ_1 (if they have above-target cash holdings) or λ_2 (if they have below-target cash holdings). The current literature on corporate capital structures suggests that costs of adjustments may prevent firms from undertaking continuous leverage adjustments (Leary and Roberts 2005; Strebulaev 2007). Borrowing that line of argument, λ_1 and λ_2 are expected to be between 0 and 1 with a higher value indicating a faster speed of adjustment. In addition, since it may be less costly for firms to disgorge than to build up cash reserves, firms with above-target cash holdings i.e., $\lambda_1 > \lambda_2$.

After classifying firms into two groups i.e., those with below- and those with abovetarget cash holdings and examining their speeds of cash holdings adjustments, the most important task is to look at how they make adjustments in each of the three cash flow categories to move toward their target cash holdings. I argue that in addition to adjustments in *CFF* which are determined by financial constraints and costs of external financing, firms can also adjust toward their target cash holdings by adjusting *CFO* and *CFI*. Those with above-target cash holdings may reduce *CFF* (e.g., to retire debt, repurchase equity, and increase dividends) and/or *CFO* (e.g., to invest more in capital expenditures and make more acquisitions), mechanisms opposite to those undertaken by firms with below-target cash holdings and these different mechanisms of adjustments may have important adjustment cost implications.

First, one would expect that it is relatively less costly for firms to retire debt, repurchase equity, and increase dividends than otherwise (Leftwich and Zmijewski 1994; Byoun 2008). Second, as investment opportunities imply potential cash flows generated in the future, opportunity costs may incur when firms have to forgo positive *NPV* investments (e.g., to decrease capital expenditures and make fewer acquisitions) due to a cash shortage and costly external financing. Minton and Schrand (1999) find that firms' liquidity constraints may force them to permanently forgo their investment opportunities rather than changing their timing.

Third, it may be easier and less costly for firms with above-target cash holdings to reduce *CFO* than for those with below-target cash holdings to increase it. In particular, the

former firms can reduce *CFO* by increasing working capital. An increase in working capital can be achieved by increasing inventories and/or accounts receivable and/or reducing accounts payable. Compared to a decrease in inventories, an increase in inventories is less likely to cause any significant impact on firms' ability to meet customers' demand, thus lowering inventory shortage costs. While costs of holding inventories can be reasonably measured, it is hard to estimate costs of lost sales due to an inventory shortage (Chiang and Monahan 2005). Hence, firms may be unwilling to reduce their inventories to avoid such costs unless they can manage inventories more effectively. Next, contrary to a fall in accounts receivable which suggests tightened credit terms, an increase in these may imply firms are extending credit to their customers, thus facilitating sales. Finally, it may be easier for firms to reduce than to increase accounts payable for the reasons discussed earlier.

3 Data, sample selection and descriptive statistics

My sample includes non-financial firms in the G-5 countries for the 1980–2007 period. Similar to Jiang and Lie (2016), I stop before the financial crisis, during which period, the determinants of cash holding are quite different from those during the non-crisis period. The sample firms' accounting data are collected from *Datastream Worldscope*. To run two-step SYS-GMM (System Generalized Method of Moments) regressions, the sample firms are required to have at least five consecutive annual observations. Those with SIC code from 6000 to 6999 (financial firms) and from 4000 to 4999 (utility firms) are excluded from the sample as they are heavily regulated and hence may exhibit atypical cash holdings behaviors. To remove outliers, I winsorize all the variables of interest at the 0.5% and 99.5% percentiles. Finally, I have a sample with 103,562 firm-year observations. Table 1 provides the statistics summary for the variables considered in the study. The size of the standard deviations indicates a reasonable level of variation in my sample which is comparable to that of previous studies.

4 Empirical results

4.1 Determinants of cash holdings and the asymmetry in speeds of adjustments

My unreported fixed-effects estimation results for the target cash holdings model are generally consistent with previous empirical evidence (e.g., Opler et al. 1999; Ozkan and Ozkan 2004; Dittmar and Duchin 2010; Venkiteshwaran 2011; Jiang and Lie 2016; Orlova and Rao 2018). In particular, in line with the precautionary motive argument and the trade-off view of cash holdings, firms' cash holdings are positively related to their growth opportunities and research and development expenses. The impact of cash flow volatility is negligible. Consistent with both the transaction cost and precautionary motive arguments and the credit line argument; larger firm size reduces firms' need to hoard cash due to the presence of economies of scale, fewer information asymmetries, and more access to bank lines of credit. There is, however, evidence that cash is positively related to cash flows which is in support of Opler et al. (1999) and Bates et al. (2009) that firms with stronger cash flows tend to accumulate more cash and may have more growth opportunities. Consistent with Fazzari et al. (1988) that financially unconstrained (possibly cash-rich) firms are likely to pay dividends, I find evidence that firms that pay dividends hoard more cash. Finally, I

| | France | | | Germany | | | Japan | | | UK | | | SU | | |
|--------------------------------------|-------------------|---------------------------|--------------|----------|------------|----------|-------------|-----------|-----------|-------------|------------|-------------|------------|-------------|------------|
| | MEA | MED | STD | MEA | MED | STD | MEA | MED | STD | MEA | MED | STD | MEA | MED | STD |
| CH _t | 0.123 | 0.086 | 0.116 | 0.121 | 0.069 | 0.143 | 0.162 | 0.133 | 0.116 | 0.122 | 0.071 | 0.151 | 0.146 | 0.071 | 0.185 |
| $BSND_t$ | 0.109 | 0.126 | 0.221 | 0.088 | 0.095 | 0.268 | 0.091 | 0.089 | 0.258 | 0.054 | 0.079 | 0.247 | 0.087 | 0.130 | 0.313 |
| ML_t | 0.338 | 0.314 | 0.232 | 0.271 | 0.225 | 0.236 | 0.347 | 0.315 | 0.261 | 0.199 | 0.155 | 0.188 | 0.243 | 0.181 | 0.233 |
| BL_t | 0.232 | 0.221 | 0.150 | 0.208 | 0.179 | 0.178 | 0.253 | 0.228 | 0.195 | 0.176 | 0.153 | 0.153 | 0.233 | 0.212 | 0.190 |
| GO_t | 0.925 | 0.734 | 0.731 | 1.028 | 0.814 | 0.877 | 0.901 | 0.742 | 0.665 | 1.316 | 0.972 | 1.293 | 1.684 | 1.111 | 1.994 |
| AT_t | 0.216 | 0.193 | 0.147 | 0.280 | 0.254 | 0.183 | 0.309 | 0.295 | 0.165 | 0.331 | 0.296 | 0.228 | 0.314 | 0.267 | 0.220 |
| FS_t | 12.053 | 11.881 | 1.974 | 12.113 | 11.885 | 1.959 | 12.872 | 12.710 | 1.547 | 10.984 | 10.810 | 1.988 | 11.885 | 11.809 | 2.061 |
| CFV_{i} | 1.144 | 0.349 | 2.056 | 1.923 | 0.557 | 3.521 | 1.569 | 0.488 | 2.777 | 2.059 | 0.608 | 4.003 | 1.856 | 0.523 | 3.819 |
| CF_t | 0.063 | 0.068 | 0.064 | 0.059 | 0.070 | 0.094 | 0.041 | 0.041 | 0.042 | 0.035 | 0.065 | 0.148 | 0.031 | 0.077 | 0.234 |
| $CAPEX_t$ | 0.055 | 0.044 | 0.047 | 0.067 | 0.051 | 0.061 | 0.038 | 0.029 | 0.035 | 0.063 | 0.047 | 0.060 | 0.068 | 0.050 | 0.066 |
| NWC_t | 0.052 | 0.050 | 0.159 | 0.113 | 0.113 | 0.193 | -0.010 | -0.005 | 0.151 | 0.034 | 0.030 | 0.181 | 0.108 | 0.109 | 0.222 |
| $R\&D_i$ | 0.009 | 0.000 | 0.023 | 0.015 | 0.000 | 0.033 | 0.013 | 0.003 | 0.020 | 0.016 | 0.000 | 0.048 | 0.053 | 0.000 | 0.154 |
| Observations | | 7477 | | | 5230 | | | 28,035 | | | 18,192 | | | 44,628 | |
| Firms | | 688 | | | 449 | | | 2960 | | | 1623 | | | 4149 | |
| This table press tions are provid | ents the de de de | scriptive st Appendix" | tatistics (m | lean—ME/ | A, median– | -MED, ar | nd standard | deviation | STD) of t | he variable | s consider | ed in the p | paper. The | se variable | s' defini- |

Table 1 Summary statistics

| Table 2 | Target cash holdings |
|----------|----------------------|
| adjustm | ents conditional on |
| deviatio | ns from target cash |
| holding | 5 |

| | France | Germany | Japan | UK | US |
|-----------------------------|--------------------|--------------------|--------------------|--------------------|--------------------|
| $CDev_{ii}$ · CH^a (i) | 0.487** (14.09) | 0.496** (13.16) | 0.469** (28.26) | 0.570** (23.24) | 0.544** (31.75) |
| $CDev_{it} \cdot CH^b$ (ii) | 0.304** (7.67) | 0.297** (5.51) | 0.318** (16.88) | 0.333** (10.76) | 0.414** (18.69) |
| Constant | 0.003** (2.70) | 0.004* (1.96) | 0.000 (-0.49) | 0.004** (4.02) | 0.002** (2.63) |
| AR2 | 0.08 | 0.40 | 0.03 | 0.23 | 0.00 |
| F-test [(i)=(ii)] | 0.00 | 0.01 | 0.00 | 0.00 | 0.00 |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 |

This table presents the SYS-GMM regression results for asymmetric partial target adjustments conditional firms' on deviations from target cash holdings, as modeled by Eq. (3): $\Delta CH_{ii} = \lambda_0 + \lambda_1 CDev_{ii}CH_{ii}^a + \lambda_2 CDev_{ii}CH_{ii}^b + \varepsilon_{ii}$, where ΔCH_{ii} is the change in cash holdings ratios. Target cash holdings are estimated by Eq. (2). See the "Appendix" for variables' definitions. $CDev_{ii}$ is the difference between cash holdings in the last period and target cash holdings for the current period. $CH_{it}^a(CH_{it}^b)$ is a dummy variable equal to 1 if cash holdings are higher than or equal to (lower than) targets and 0 otherwise. My sample includes 103,562 firm-year observations for France, Germany, Japan, the UK and the US over the 1980-2007 period. Figures in parentheses are t-statistics. F-test reports the *p*-value of the *F*-test for the hypothesis that the estimated coefficients of speeds of adjustments for firms with above- and those with belowtarget cash holdings are equal. AR2 reports the p-value of the test for no second-order serial correlation, which is asymptotically distributed as N(0,1) under the null hypothesis of no serial correlation. ** and * indicate that the estimated coefficients are significant at the 1, and 5% levels of significance, respectively

show that both capital expenditures and net working capital have a significant, negative impact on cash holdings. This finding is consistent with Bates et al. (2009) and Riddick and Whited (2009) that significant investments in assets may temporarily reduce firms' cash holdings and net working capital may reduce firms' need to hold cash since it contains highly liquid assets which can be considered as close substitutes for cash.

Table 2 reports the estimation results for the asymmetric partial adjustment model as specified by Eq. (3). These results strongly support the trade-off view of corporate cash holdings that there is asymmetry in firms' speeds of adjustments. Conditional on having above-target cash holdings, firms adjust toward their target cash holdings at speeds which are both economically and statistically faster than speeds of adjustments of firms with below-target cash holdings, as indicated by the *F*-tests. Such evidence would suggest that on average, speeds of adjustments for firms in the five sample countries may be 13-24% faster when they have above- than when they have below-target cash holdings.

4.2 Main results—the asymmetry in mechanisms of cash holdings adjustments

4.2.1 Mechanisms of adjustments among firms with above-target cash holdings

Table 3 provides detailed explanations on why firms with above-target cash holdings may adjust toward their targets faster. There is evidence that these firms reduce *CFF* (across

| | France | | German | ny | Japan | | UK | | US | |
|----------|------------|-------------|-------------|---------|-------|------|------|------|------|------|
| | % | Rank | % | Rank | % | Rank | % | Rank | % | Rank |
| Panel A: | firms with | h above-tai | rget cash h | oldings | | | | | | |
| CFF | -1.4 | [1] | -1.1 | [2] | -0.8 | [2] | -2.8 | [1] | -2.3 | [1] |
| CFO | -0.6 | [3] | -0.1 | [3] | -0.9 | [1] | +0.1 | [3] | 0.0 | [3] |
| CFI | +1.3 | [2] | +1.9 | [1] | +0.7 | [3] | +2.3 | [2] | +1.6 | [2] |
| Panel B: | firms with | n below-tai | rget cash h | oldings | | | | | | |
| CFF | +0.8 | [3] | +0.3 | [3] | +0.7 | [2] | +0.6 | [3] | +1.8 | [2] |
| CFO | +1.5 | [1] | +2.0 | [1] | +1.3 | [1] | +2.0 | [2] | +2.3 | [1] |
| CFI | -1.1 | [2] | -1.8 | [2] | -0.2 | [3] | -2.2 | [1] | 0.0 | [3] |

 Table 3
 Magnitudes of cash flow adjustments conditional on deviations from target cash holdings

This table represents the magnitudes of the adjustments in *CFF*, *CFO*, and *CFI* which are defined as changes in *CFF*, *CFO*, and *CFI* scaled by firms' total assets conditional on their deviations from target cash holdings together with their relative ranks ranging from to 1 (largest) to 3 (smallest). See the "Appendix" for variables' definitions

firms in all countries) and *CFO* (except for UK firms which experience a very slight increase and US firms which see no change in *CFO*) but increase *CFI* (across firms in all countries). These mechanisms of adjustments are likely to involve lower costs of adjustments, thus allowing these firms to adjust toward their targets faster to reduce potential costs associated with the free cash flow problem. In contrast, firms with below-target cash holdings undertake opposite mechanisms (to increase *CFF* and *CFO* (across firms in all countries) but reduce *CFI* (except for US firms which see no change in *CFI*). The magnitudes of the adjustments in each cash flow group vary significantly between these two groups of firms and will be discussed shortly.

I find that overall, firms with above-target cash holdings experience a fall in their cash balances, as can be seen from Panel A in Table 3. Be aware that Panel A of Table 3 reports the magnitudes and relative ranks of the adjustments in the three cash flow groups (*CFF*, *CFO*, and *CFI*) among firms with above-target cash holdings. To make it easier to compare the magnitudes of firms' cash holdings adjustments, the changes in *CFF*, *CFO*, and *CFI* are taken from Panel A in Table 4 to Panel A of Table 3.

Adjustments in CFF Adjustments in CFF are significant across firms in all the sample countries (ranked 1st among French, UK, and US firms with 1.4%, 2.8%, and 2.3% of total assets in monetary terms, respectively) (Panel A of Table 3), especially UK and US firms for the magnitude of the adjustments in this cash flow group is much greater than that of the others.³ As can be seen from Panel A of Table 4, these firms consistently experience a fall in their levels of net equity issues (0.3%, 0.7%, 0.4%, 3.1%, and 2.6% of total assets in France, Germany, Japan, the UK, and the US, respectively), which is mainly driven by the decrease in their levels of equity proceeds. In addition, French, German, and Japanese firms also see a decline in their levels of net debt issues. These together with a slight increase in levels of dividend payout then lead to a significant decrease in *CFF*.

³ The adjustments in net equity issues among UK and US firms are more significant than those of firms in France, Germany, and France, suggesting these firms may be relatively more active in the equity market. Indeed, Öztekin (2015) shows that in countries with stronger shareholder rights and better enforcement of these rights, firms are subject to lower equity costs and have increased recourse to equity financing.

Table 4 Firms' cash flow adjustments conditional on their deviations from target cash holdings

- 17.89 -41.96-32.18-10.90-20.13-25.85-13.7410.79 67.90 161.69 35.82 55.38 -2.48-13.08-10.86 - 17.75 -18.594.98 6.79 2.88 8.04 4.8 -4.47 S 26.85 -14.34- 15.57 22.73 -3.48 -6.68 -7.43 4.06 10.37 -10.85- 10.89 - 11.85 5.2646.04 96.23 24.50 34.26 12.34 5.40-4.57-6.71 8.40 4.63 UΚ -2.19- 12.57 23.64 -8.47-1.90 51.76 42.78 - 39.94 -30.68-8.92 3.39 -7.52 -5.26-8.29 0.48 -8.51 29.39 55.79 -7.41 9.93 0.73 7.51 4.75 JPA -14.5512.23 - 1.66 -1.55-4.06-8.56 - 7.67 - 7.93 1.79 23.45 52.79 13.52 19.26 -4.33 -4.87 3.29 3.32 -8.31- 2.21 4.33 7.36 0.84 2.75 GER -13.29 -1.10-3.38-3.25-4.73-1.0813.35 - 16.41 -4.64 -4.93 -3.75 -6.58 -3.52 -2.5357.53 22.60 2.036.65 2.89 -3.912.01 0.40 23.71 t-stats FRA -0.052-0.004-0.0240.014 0.2460.0890.118 0.028 0.026 0.018 0.049 0.036 0.016 0.011 0.060 0.023 0.072 0.000 0.017 0.013 0.2500.037 0.001 SD -0.022-0.051-0.0240.043 0.016 0.015 0.000 0.073 0.099 0.024 0.025 0.018 0.006 0.060 0.020 0.055 0.046 0.003 0.011 0.202 0.202 0.000 0.011 UK Panel B-firms with below-target -0.002-0.032-0.020-0.011-0.006 -0.000 0.118 0.2620.002 0.1300.012 0.012 0.0070.054 0.013 0.007 0.002 0.005 0.002 0.251 0.001 0.031 0.001 JPA -0.018-0.042-0.001- 0.001 -0.0210.075 0.096 0.019 0.015 0.012 0.015 0.249 0.020 0.003 0.020 0.053 0.004 0.249 0.017 0.072 0.004 0.003 0.000 GER Cash holdings -0.038-0.011-0.0210.015 090.0 0.016 0.014 0.003 0.2490.250 0.085 0.0200.022 0.009 0.008 0.072 D.004 0.002 0.001 0.001 0.001 0.101 0.001 FRA -0.016-0.026-0.020-0.023-0.019-0.011-0.0210.0300.214 0.010 0.029 0.016 0.027 0.205 0.2340.063 0.198 0.029 0.000 0.080 0.016 0.043 0.005 SD -0.013-0.032-0.028-0.007-0.034-0.027-0.0310.165 0.005 0.030 0.1550.010 0.215 0.035 0.067 0.177 0.0320.037 0.023 0.005 0.004 0.077 0.001 Panel A-firms with above-target UK -0.018-0.004-0.010-0.012-0.010-0.008-0.009-0.008-0.0030.046 0.000 0.228 0.016 0.038 0.005 0.005 0.238 0.002 0.200 0.037 0.177 0.007 0.001 JPA -0.011-0.007-0.012-0.025-0.008-0.010-0.027-0.001-0.0050.009 0.192 0.1960.005 0.056 0.053 0.069 0.019 0.013 0.004 0.204 0.1740.027 0.002 GER Cash holdings -0.007-0.014-0.007-0.012-0.013-0.013-0.006 -0.001-0.003-0.001-0.0040.059 0.230 0.004 0.008 0.228 0.065 0.013 0.012 0.177 0.044 0.1550.023 FRA Equity proceeds change, Cash & S-T investment Equity repurchases, Net equity change_t Equity repurchase Equity proceeds $_t$ CFF change_t CFO change_t CFI change_t CH change, $BL \ change_t$ CHDev_{i-1} Net equity_t BLDev_{t-1} $change_t$ CHDev_t $change_t$ $BLDev_{t}$ CH_{t-l} BL_{t-1} CFO_{t} CFF, CFI, CH_t BL_t

| \sim |
|--------|
| [pai |
| tint |
| con |
| C |
| 4 |
| ه |
| 9 |
| Ъ |

| | Panel A- Cash holo | firms wi dings | th above-t | arget | | Panel B– Cash hold | -firms wit lings | th below-tar | get | 1 | t-stats | | | | |
|--|-----------------------|-------------------|------------|--------------------|--------|-----------------------|---------------------|---------------------|--------------------|--------|---------|--------|--------|-------|---------|
| | FRA | GER | JPA | UK | NS | FRA | GER | JPA | UK | I SU | FRA | GER | IPA | UK | SU |
| Net $debt_t$ | -0.003 | -0.002 | -0.011 | -0.002 | 0.013 | 0.006 | - 0.001 | - 0.007 | -0.003 | 0.010 | -2.55 | - 0.29 | -2.88 | 3.11 | 3.07 |
| Net debt change _t | -0.009 | -0.002 | -0.003 | 0.005 | 0.005 | 0.006 | 0.000 | 0.006 | -0.004 | 0.003 | -1.04 | -0.31 | -5.80 | 3.63 | 2.45 |
| Cash dividends _t | 0.012 | 0.015 | 0.007 | 0.023 | 0.010 | 0.011 | 0.012 | 0.006 | 0.022 | 0.009 | 0.41 | 4.57 | 6.45 | 1.19 | 1.49 |
| Cash dividend change _t | 0.002 | 0.002 | 0.001 | 0.002 | 0.002 | 0.001 | 0.001 | 0.000 | 0.001 | 0.001 | 1.40 | 2.89 | 7.73 | 4.71 | 3.25 |
| Funds from operations _t | 0.064 | 0.065 | 0.053 | 0.045 | 0.041 | 0.067 | 0.072 | 0.052 | 0.062 | 0.067 | -0.90 | - 1.63 | 1.29 | -4.74 | - 8.38 |
| Funds from operations change _t | 0.006 | 0.013 | 0.001 | 0.009 | 0.008 | 0.005 | 0.013 | 0.007 | 0.014 | 0.017 | 0.01 | 0.61 | - 1.89 | -1.94 | - 3.88 |
| Working capital _t | -0.005 | -0.011 | -0.006 | -0.008 | -0.011 | 0.004 | -0.001 | 2×10^{-4} | -0.002 | -0.007 | -4.12 | - 3.37 | -9.84 | -4.04 | -4.07 |
| Working capital change _t | 0.012 | 0.014 | 0.010 | 0.008 | 0.008 | -0.010 | -0.007 | -0.006 | - 0.006 | -0.006 | 7.24 | 4.86 | 17.88 | 6.33 | 8.57 |
| Net assets from acqui- sitions _t | 0.007 | 0.017 | 0.002 | 0.038 | 0.025 | 0.005 | 0.012 | 0.002 | 0.031 | 0.021 | 1.68 | 2.47 | 2.35 | 4.41 | 5.96 |
| Change in net assets from acquisitions _t | 0.001 | 0.004 | 0.001 | 0.010 | 0.010 | -0.002 | -0.004 | 0.000 | - 0.006 | -0.003 | 3.10 | 3.60 | 4.71 | 7.76 | 14.51 |
| Increase in investments $_{t}$ | 0.018 | 0.024 | 0.035 | 0.008 | 0.100 | 0.016 | 0.015 | 0.025 | 0.006 | 0.050 | 1.46 | 4.00 | 9.64 | 2.29 | 20.38 |
| Change in increase in investments _t | 0.000 | 0.001 | 0.001 | 0.000 | 0.002 | -0.002 | - 0.003 | -0.003 | 0.000 | 0.007 | 1.05 | 1.38 | 4.62 | -0.06 | - 2.47 |
| Decrease in invest- ments _t | 0.010 | 0.024 | 0.033 | 0.007 | 0.104 | 0.011 | 0.015 | 0.026 | 0.007 | 0.047 | -0.60 | 3.42 | 7.00 | 1.15 | 22.48 |
| Change in decrease in investments _t | -0.004 | - 0.006 | 0.000 | -0.001 | 0.010 | 0.004 | 0.001 | -0.003 | 0.001 | 0.000 | -3.74 | - 2.01 | 3.37 | -2.70 | 5.74 |
| Net investments $_t$ | 0.009 | 0.001 | 0.002 | 0.001 | -0.002 | 0.006 | 0.001 | -0.001 | 4×10^{-4} | 0.004 | 1.50 | 0.30 | 5.96 | 0.42 | -5.73 |
| Change in net invest- ments _t | 0.004 | 0.006 | 0.001 | 3×10^{-5} | -0.008 | -0.005 | - 0.003 | -1×10^{-5} | - 0.001 | 0.007 | 3.43 | 2.20 | 1.64 | 1.95 | - 10.46 |
| Fixed asset disposal _t | 0.011 | 0.013 | 0.006 | 0.023 | 0.011 | 0.012 | 0.022 | 0.008 | 0.032 | 0.014 | -1.12 | -5.31 | -6.18 | -6.84 | -6.12 |

| - | |
|------|---|
| · C | |
| - 6 | 1 |
| 2 | - |
| - 5 | |
| - 2 | _ |
| • 7 | |
| - * | = |
| - 1- | |
| - C | 2 |
| ē | 5 |
| ~ | 1 |
| | |
| | |
| ~ | r |
| - 2 | 1 |
| - Q | U |
| - | |
| _ | 1 |
| - | |
| ÷ | 2 |
| | |

| | Panel A- Cash hol | -firms wi dings | th above-t | arget | | Panel B– Cash holo | -firms wit lings | h below-ta | rget | | t-stats | | | | |
|--|----------------------|--------------------|------------|------------|-----------|-----------------------|---------------------|-------------|-------------|------------|----------|------------|-----------|------------|-----------|
| | FRA | GER | JPA | UK | SU | FRA | GER | JPA | UK | SU | FRA | GER | JPA | UK | SU |
| Change in fixed asset disposal _t | -0.002 | - 0.004 | - 0.001 | -0.007 | -0.002 | 0.002 | 0.004 | 0.001 | 0.010 | 0.003 | -4.21 | - 4.15 | -5.68 | -9.76 | - 9.49 |
| Other uses/(sources) of investing _t | 0.007 | 0.003 | 0.002 | 0.000 | -0.001 | 0.005 | 0.001 | 0.001 | 0.000 | -0.003 | 1.75 | 1.22 | 4.25 | 1.36 | 3.66 |
| Change in other uses/ (sources) of investing _t | 0.002 | 0.002 | 0.001 | 0.000 | 0.006 | -0.002 | - 0.002 | 0.000 | 0.000 | -0.004 | 2.47 | 2.81 | 4.13 | 1.39 | 6.01 |
| $CAPEX_t$ | 0.048 | 0.053 | 0.036 | 0.055 | 0.065 | 0.050 | 0.055 | 0.035 | 0.052 | 0.061 | -1.05 | - 1.21 | 0.67 | 2.61 | 4.66 |
| CAPEX change _t | 0.001 | -0.001 | 0.002 | 0.002 | 0.005 | 0.001 | -0.002 | 0.001 | 0.000 | 0.003 | 0.55 | 0.66 | 2.78 | 2.50 | 3.47 |
| Observations | 3906 | 2602 | 15,230 | 9198 | 22,645 | 3571 | 2628 | 12,805 | 8994 | 21,983 | | | | | |
| This table summarizes t | he mean le | vels of fir | ms' cash | flow adine | stments P | anel A is | for firms | with above. | -taroet cas | h holdinos | and Pane | I R is for | firms wif | h helow-ta | roet cash |

holdings in the last accounting period in five countries—France (FRA), Germany (GER), Japan (JPA), the UK (UK), and the US. US. See the "Appendix" for variables' definitions. *I*-statistics report the *i*-values of the *t*-test for the hypothesis that the means for firms with above-target cash holdings and those with below-target cash holdings in each country are equal Compared to their counterparts in France, Germany and Japan, UK and US firms reduce their levels of net equity issues very significantly (3.1% and 2.6% of total assets in monetary terms, respectively) and still increase their levels of net debt issues (0.5%) (Panel A of Table 4).⁴ The net effect of these adjustments is therefore a fall in *CFF*. Interestingly, by magnitude, the increase in levels of dividend payout is smaller than the increase in levels of equity repurchases among UK and US firms (0.2% in each country). This finding is consistent with managerial perception and empirical evidence that firms' managers prefer to distribute some of their firms' excess cash in the way that is likely to establish less commitment (Iyer et al. 2017).

The reason why managers prefer to distribute their firms' excess cash to shareholders in the form of share repurchases rather than dividend increases has been discussed intensively. The current literature on firms' dividend policies since Miller and Modigliani (1961) suggests that investors tend to interpret firms' dividend changes as an indicator of changes in their management's view about their firms' future prospects (Bhattacharya 1979; John and Williams 1985; Miller and Rock 1985; Best and Best 2001). An increase in dividends may indicate that firms' managers are optimistic about their firms' future while a fall in these implies the opposite.⁵ Dividend cuts therefore may be considered as a "last resort" action for firms and investors tend to associate these with financial problems that may not reverse in the near term (Teng and Hachiya 2013). Leftwich and Zmijewski (1994) show that a dividend reduction has much more information content about changes in firms' future operations than a dividend increase as it signals serious deterioration in firms' longterm prospects. Hence, managers tend to be unwilling to increase dividend payments by large amounts as once they have done so, it may be tough for them to cut these latter during times of adverse cash flow shocks. Equity repurchases therefore may be a much more flexible approach to distribute firms' excess cash to shareholders from managers' point of view.⁶

The evidence on the small change in firms' dividend payout whether they experience above- or below-target cash holdings is particularly interesting. Apart from its implication about managers' incentives to distribute their firms' excess cash in the way that is likely to establish a lower level of commitment, it may also reflect firms' attempts to smooth dividends which have been acknowledged by a vast body of the current literature on corporate dividend policies. For example, Brav et al. (2005) find that as managers tend to believe that the market is likely to put a premium on firms that have stable dividend policies due to the signaling effect of any negative changes on these, they are willing to visit costly capital markets or even give up positive *NPV* investments rather than cutting their dividends.⁷

⁴ Lee and Suh (2011) find that the increase in cash holdings prior to share repurchases is obtained from the decrease in firms' capital expenditures rather than the improvement in their operating performance.

⁵ The literature review of Allen and Michaely (2003), however, shows that the overall accumulated empirical evidence does not support the assertion of traditional dividend signaling models that dividend changes convey information about firms' future earnings as contrary to the signaling theory's prediction; it indicates that there is no positive link between dividend changes and future earnings.

⁶ Another potential reason is that realizing capital gains is generally more tax efficient for shareholders compared to dividends.

⁷ Firms' incentives for dividend smoothing can be explained by information asymmetry models (e.g., coarse signaling models (Kumar 1988; Kumar and Lee 2001; Guttman et al. 2010), principal-agent models (Fudenberg and Tirole 1995; DeMarzo and Sannikov 2008), information asymmetry among investors (Brennan and Thakor 1990) and external financial constraints (Almeida et al. 2005; Bates et al. 2009), agency-based models (Jensen 1986; Allen et al. 2002; DeAngelo and DeAngelo 2007; Lambrecht and Myers 2010), and income smoothing models (Miller and Scholes 1978; Baker et al. 2007; Baker and Wurgler 2010).

Adjustments in CFI Contingent on having above-target cash holdings, French and German firms adjust CFI mainly through adjustments in their portfolio investments, short-term investments, and marketable securities (Panel A of Table 4). On the contrary, as the mergers and acquisitions (M&A) markets may be more active in the UK and the US, firms in these two countries experience a significant rise in their levels of net assets from acquisitions (1.0%) of total assets in monetary terms in each country). The significant increase in levels of net assets from acquisitions among UK and US firms is in support of Harford (2002) and Harford et al. (2012) that cash-rich firms are more likely to make acquisitions even when their acquisitions may be value-decreasing as their managers tend to enjoy an increase in their compensation following these acquisitions (Bliss and Rosen 2001; Harford and Li 2007). Contingent on having above-target cash holdings, firms see a decrease in their levels of fixed asset disposal. Finally, US firms with above-target cash holdings experience the most significant increase in capital expenditures among firms in the five sample countries (0.5%). This finding is consistent with the statistics reported in Table 1 that these firms on average have most growth opportunities, as suggested by their market-to-book ratios.

Adjustments in CFO Firms with above-target cash holdings in Germany and the UK experience a small change in CFO and those in the US do not see any change in it (smallest among the three cash flow groups) (Panel A of Table 3) as the increase in their levels of funds from operations closely matches with the rise in their levels of working capital (Panel A of Table 4). In contrast, the change in CFO is rather large for French and Japanese firms (0.6% and 0.9%, respectively) as their levels of working capital increase at higher rates than levels of funds from operations. The size mismatch between the rise in levels of working capital and that in levels of funds from operations among French and Japanese firms supports the view that in the presence of weak corporate governance, excess cash can be used unproductively (Dittmar and Mahrt-Smith 2007). Indeed, Ando et al. (2003) find that corporate governance among Japanese firms overall is weak and ineffective.

4.2.2 Mechanisms of adjustments among firms with below-target cash holdings

Next, I examine the mechanisms undertaken by firms with below-target cash holdings to move toward their targets. I find that these firms try to increase their cash balances, as can be seen from Panel B in Table 4. Further, there is evidence that overall firms with below-target cash holdings undertake mechanisms of adjustments opposite to those undertaken by firms with above-target cash holdings. Panel B of Table 3 reports the magnitudes and relative ranks of their adjustments in the three cash flow groups. The magnitudes of these adjustments are taken from Panel B in Table 4.

Adjustments in CFF I find that firms with below-target cash holdings adjust toward their targets by increasing CFF. However, since these firms are over-levered in the previous accounting period $(BLDev_{t-1} > 0)$, as shown in Panel B of Table 4, it may be costly for them to visit external capital markets. This is probably why the magnitude of the increase in CFF is particularly small among German, Japanese, and UK firms (smallest among the three cash flow groups) (Panel B of Table 3). Due to their significant deviations from target leverage, German firms do not experience any change in their levels of net debt issues

while UK firms even have to reduce them to avoid potential financial distress (Panel B of Table 4).⁸

Adjustments in CFO The change in firms' CFO becomes an important driver of their cash holdings adjustments (ranked 1st in most countries) (Panel B of Table 3) in the presence of below-target cash holdings as it may be more costly for these firms to visit capital markets. The increase in levels of funds from operations among Japanese, UK, and US firms is greater when they experience below-target than when they have above-target cash holdings. In addition, there is evidence that the magnitude of the fall in levels of working capital among firms with below-target cash holdings is much smaller than that of the increase in levels of funds from operations (except for French firms where the change in levels of working capital is more significant and Japanese firms which see almost equal changes in these two items) (Panel B of Table 4), suggesting a significant improvement in their levels of operational efficiency.

Even when firms with below-target cash holdings need to increase cash, they may be still concerned about potential impacts of a significant reduction in their working capital on their operations as the magnitude of the decrease in their levels of working capital is smaller than that of the increase in levels of working capital among firms with above-target cash holdings. The implication here is, when these firms have to reduce working capital to improve *CFO* (e.g., to reduce inventories and accounts receivable and/or increase accounts payable), they still need to ensure their ability to meet their customers' demand and offer them reasonable credit terms and meet their suppliers' credit requirements.

Adjustments in CFI The presence of below-target cash holdings forces firms, especially those which are highly over-levered and hence may have less access to capital markets (e.g., German and UK firms) to reduce CFI (except for US firms) (Panel B of Table 3). The decrease in levels of net assets from acquisitions among German and UK firms is particularly significant. The noticeable rise in German and UK firms' levels of fixed asset disposal (Panel B of Table 4) may imply their attempts to boost operational efficiency i.e., to streamline their operations by selling less productive assets to improve both profitability and liquidity.

Overall, in support of the trade-off view, I find evidence that firms with above-target cash holdings experience faster speeds of adjustments since their mechanisms of adjustments may incur lower costs than those undertaken by firms with below-target cash holdings. These firms adjust toward their targets generally through significant changes in *CFF* (e.g., they generally decrease their levels of equity proceeds (all firms) and net debt issues (except for UK and US firms) but increase their levels of equity repurchases (except for French firms) and dividend payout (all firms)) and *CFI* (e.g., they generally increase their levels of net assets from acquisitions (all firms), portfolio and short-term investments (except for US firms), and capital expenditures (except for German firms) but reduce their levels of fixed asset disposal (all firms)) while firms with below-target cash holdings adjust mainly through changes in *CFO* i.e., to increase their levels of funds from operations but decrease their levels of working capital (all firms). The evidence on the asymmetry in the adjustments in *CFO* and *CFI* undertaken by these two groups of firms is especially novel. Previous studies tend to treat these adjustments as exogenous, independent of firms' cash

⁸ I find some link between cash holdings and leverage adjustments among firms with below-target cash holdings. Over-levered firms with large deviations from target leverage may find it harder to increase levels of net debt issues. Such a link is not found among those with above-target cash holdings.

and leverage management policies but I now explicitly account for these to get a better insight into firms' cash holdings adjustments.

5 Additional analyses

5.1 The asymmetry in the impact of the magnitude of deviations from targets on cash holdings adjustments

The magnitude of firms' deviations from target cash holdings is likely to shape their target adjustments for two reasons. First, considering the magnitude of firms' cash flow realizations, Faulkender et al. (2012) realize that the presence of large operating cash flows may encourage firms to adjust toward their target leverage faster as costs of leverage adjustments can be effectively shared with transaction costs that incur when they offset these cash flow imbalances, thus lowering the marginal costs of leverage adjustments. That line of argument can be borrowed to explain firms' target cash holdings adjustments contingent on the magnitude of their deviations from target cash holdings. In particular, I argue that the presence of the fixed component of costs of adjustments (e.g., fixed costs of debt/equity issues, permanent losses of customers and growth opportunities, etc.) may discourage firms with small deviations to rebalance their cash. Large deviations, however, may lead to lower incremental costs of adjustments as the fixed cost component can now be effectively "shared" with transaction costs that incur when firms offset their cash holdings imbalances and hence faster speeds of adjustments.

Second, agency theories argue that firms with more free cash are more likely to experience the free cash flow problem (Jensen 1986), thus having more incentives to reduce cash to mitigate the potential costs associated with it. Firms with a large cash shortage, however, are likely to have more pressures to adjust to avoid suffering from operational problems and having to give up positive *NPV* investments. These costs of deviations, however, may be lower for firms with little excess cash (as managers do not have much to squander) or a small cash shortage (as the probability of experiencing operational problems and having to give up positive *NPV* projects is lower).

To examine the impact of the magnitude of firms' deviations from target cash holdings on their target adjustments, I divide both firms with above-target cash holdings and those with below-target cash holdings into two subgroups using the median level of deviations from target cash holdings for each group. My approach here extends that of Dittmar and Duchin (2010) as I take into account whether firms have above- or below-target cash holdings while the authors do not. Firms with large (small) deviations may adjust more quickly (slowly) for different reasons contingent on whether they experience above or below-target cash holdings.

$$\Delta CH_{it} = \lambda_0 + \left(\lambda_1 CDev_{it}^L + \lambda_2 CDev_{it}^S\right) CDev_{it}CH_{it}^a + \left(\lambda_3 CDev_{it}^L + \lambda_4 CDev_{it}^S\right) CDev_{it}CH_{it}^b + \varepsilon_{it}.$$
(4)

Here $CH_{ii}^{a}CDev_{ii}^{L}$ $(CH_{ii}^{a}CDev_{ii}^{S})$ indicates firms experience above-target cash holdings and the magnitude of the deviations is larger than or equal to (smaller than) the median level for their group while $CH_{ii}^{b}CDev_{ii}^{L}$ $(CH_{ii}^{b}CDev_{ii}^{S})$ indicates firms have below-target cash holdings and the magnitude of the deviations is larger than or equal to (smaller than) the median level for their group. λ_{1} may be greater than λ_{2} and λ_{3} while λ_{2} and λ_{3} may be greater than λ_{4} .

| | France | Germany | Japan | UK | US |
|---|--------------------|--------------------|--------------------|--------------------|--------------------|
| $CDev_{it}$ · CH^a_{it} · $CDev^L_{it}$ (i) | 0.502** (15.43) | 0.510** (14.19) | 0.467** (28.78) | 0.576** (23.50) | 0.548** (35.32) |
| $CDev_{it} \cdot CH^a_{it} \cdot CDev^S_{it}$ (ii) | 0.494** (6.15) | 0.455** (4.93) | 0.548** (13.65) | 0.537** (7.96) | 0.400** (8.92) |
| $CDev_{it} \cdot CH^b_{it} \cdot CDev^L_{it}$ (iii) | 0.319** (8.35) | 0.297** (5.91) | 0.317** (17.27) | 0.342** (11.16) | 0.429** (20.24) |
| $CDev_{it} \cdot CH^b_{it} \cdot CDev^S_{it}$ (iv) | 0.146 (1.86) | 0.291** (2.72) | 0.335** (7.44) | 0.288** (4.62) | 0.358** (8.46) |
| Constant | 0.004** (3.32) | 0.004* (2.17) | 0.000 (-0.27) | 0.005** (3.95) | 0.002* (2.04) |
| AR2 | 0.09 | 0.40 | 0.03 | 0.24 | 0.00 |
| F-test [(i)=(ii)] | 0.92 | 0.54 | 0.03 | 0.55 | 0.00 |
| F-test [(iii)=(iv)] | 0.04 | 0.95 | 0.69 | 0.34 | 0.08 |
| F-test [(i)=(iii)] | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 |
| F-test [(ii)=(iv)] | 0.00 | 0.28 | 0.00 | 0.02 | 0.54 |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 |

 Table 5
 Target cash holdings adjustments conditional on the magnitude of deviations from target cash holdings

This table presents the SYS-GMM regression results for firms' asymmetric partial target adjustments conditional on the magnitude of deviations from target cash holdings, as modeled by Eq. (4): $\Delta CH_{ii} = \lambda_0 + (\lambda_1 CDev_{ii}^L + \lambda_2 CDev_{ii}^S)CDev_{ii} CH_{ii}^a + (\lambda_3 CDev_{ii}^L + \lambda_4 CDev_{ii}^S)CDev_{ii} CH_{ii}^b + \epsilon_{ii}$, where ΔCH_{ii} is the change in cash holdings ratios. Target cash holdings are estimated by Eq. (2). See the "Appendix" for variables' definitions. $CDev_{ii}$ is the difference between cash holdings in the last period and target cash holdings for the current period. $CH_{ii}^a (CH_{ii}^b)$ is a dummy variable equal to 1 if cash holdings are higher than or equal to (lower than) targets and 0 otherwise. $CDev_{ii}^L (CDev_{ii}^S)$ is a dummy variable equal to 1 if deviations from target cash holdings are larger than or equal to (smaller than) the median level and 0 otherwise. My sample includes 103,562 firm-year observations for France, Germany, Japan, the UK and the US over the 1980–2007 period. Figures in parentheses are *t*-statistics. *F*-test reports the *p*-value of the *F*-test for the hypothesis that the coefficient estimates for each pair of scenarios are equal. *AR2* reports the *p*-value of the test for no second-order serial correlation, which is asymptotically distributed as N(0,1) under the null hypothesis of no serial correlation. ** and * indicate that the estimated coefficients are significant at the 1, and 5% levels of significance, respectively

The impact of the magnitude of deviations from targets is reported in Table 5. I find some evidence that firms' speeds of adjustments are also determined by how far they are away from their targets. The presence of the fixed cost component discourages US firms with small excess cash to adjust, thus leading to statistically slower speeds of adjustments compared to those of firms that have a lot of excess cash in the country. It is, however, an opposite story with Japanese firms as those with small excess cash adjust statistically faster, which is rather puzzling. Although speeds of adjustments for firms with above-target cash holdings and large deviations in France, Germany, and the UK are not statistically faster than those of firms with above-target cash holdings and small deviations, by magnitude, the former firms adjust faster, a finding to some extent in support of the "shared" fixed cost argument and agency theories on costs of deviations from target cash holdings i.e., firms with large excess cash holdings may face lower costs of adjustments but higher costs of deviations associated with the free cash flow problem.

Among firms with below-target cash holdings, although speeds of adjustments do not statistically differ between firms with large deviations and those with small deviations, by magnitude, the presence of large deviations still leads to faster speeds of adjustments (except for Japanese firms). This evidence lends some support to the earlier argument that firms with a large cash shortage may need to adjust more quickly for they are more likely to suffer from operational problems and have to give up positive *NPV* investments when costs of adjustments can be effectively "shared".

I find that conditional on having large deviations from targets, firms with above-target cash holdings adjust statistically faster than those with below-target cash holdings. A similar pattern can be also found among firms with small deviations from targets i.e., firms with small excess cash adjust statistically faster than those with a small cash shortage. These findings suggest deviations from targets i.e., whether firms have above- or below-target cash holdings may be the major driver of their target adjustments, not the magnitude of these deviations.

Firms' characteristics and actual cash holdings adjustments contingent on whether they have above- or below-target cash holdings and the magnitude of their deviations from targets reported in Table 6 can effectively shed light on why firms with large deviations from targets are likely to adjust faster than those with small ones. First, as expected, on average, firms with large deviations from targets, whether having above- or below-target cash holdings, in terms of absolute values, experience much more significant deviations from targets (from 2.8% (firms with large excess cash holdings or A-L firms in Japan) to 5.0% (A-L firms in the UK) of total assets) than those of firms with small deviations (from 0.5% (firms with small excess cash holdings or A-S firms in Japan and firms with a small cash shortage or B-S firms in all the sample countries with the exception of Japan) to 1.2% (A-S firms in the UK)). Hence, both the "shared" fixed cost argument and agency theories suggest they should undertake faster adjustments.

Second, except for French firms with above-target cash holdings and small deviations from targets, whether having above- or below-target cash holdings, firms with small deviations generally have stayed close to their target levels of leverage (especially US firms) (small $BLDev_t$). This finding implies that the presence of fixed costs related to leverage adjustments may discourage these firms to visit external capital markets to rebalance their cash holdings via adjustments in *CFF* (Faulkender et al. 2012), which explains why compared to firms with large deviations, those with small ones experience much smaller changes in *CFF* in most cases.

In line with the view that excess cash tends to be used unproductively (Dittmar and Mahrt-Smith 2007), I find that the rise in A-L firms' levels of working capital is much more significant than that in their levels of funds from operations as these firms collate a lot more accounts receivable and inventories while the improvement in their funds from operations is rather modest. A-L firms in Japan even experience a decline in their levels of funds from operations. However, when the magnitude of their excess cash holdings is small, they start trying to become more efficient as the growth of A-S firms' levels of funds from operations starts exceeding that of their levels of working capital (except for A-S firms in Germany). Especially, I find that among the four groups of firms, B-L firms or those facing a large cash shortage across all the sample countries experience the largest improvement in their levels of funds from operation and the most significant reduction in their levels of working capital. Taken together, these findings imply that firms' levels of operational efficiency vary conditional on the position relative to their target cash holdings.

Overall, I find evidence that consistent with the "shared" fixed cost argument and agency theories, firms with large deviations from target cash holdings adjust toward their target cash holdings faster than those with small deviations. Firms with small deviations may be discouraged to undertake adjustments possibly due to higher incremental costs of adjustments and lower costs of deviations. In addition, firms with a lot of excess cash do

| holdings |
|----------------|
| m target cash |
| ons fro |
| deviati |
| of |
| magnitude |
| on the |
| conditional of |
| flows |
| cash |
| п. |
| Changes |
| Table 6 |

| | A-L | A-S | B-L | B-S | t-stats | | | | | |
|---|---------|--------|--------|--------|---------|---------|---------|---------|---------|---------|
| | (1) | (2) | (3) | (4) | (1)–(2) | (1)–(3) | (1)-(4) | (2)–(3) | (2)–(4) | (3)–(4) |
| France | | | | | | | | | | |
| BL_t | 0.203 | 0.253 | 0.234 | 0.268 | -6.02 | -3.70 | -8.00 | 2.49 | -2.04 | -4.56 |
| $BLDev_t$ | -0.012 | -0.012 | 0.004 | -0.002 | -0.13 | - 4.01 | - 2.54 | -4.10 | - 2.57 | 1.68 |
| CH_{t} | 0.206 | 0.105 | 0.117 | 0.084 | 15.42 | 13.87 | 20.04 | -2.65 | 5.20 | 8.04 |
| CHD ev _t | 0.040 | 0.006 | -0.036 | -0.005 | 11.76 | 23.58 | 16.56 | 16.75 | 6.30 | -12.95 |
| $CH \ change_t$ | -0.026 | -0.001 | 0.030 | 0.009 | -7.66 | -16.47 | -11.63 | -12.00 | -5.07 | 8.77 |
| CFF_t | -0.005 | -0.008 | 0.008 | 0.010 | 0.67 | -2.49 | -3.03 | -3.54 | -4.35 | -0.35 |
| CFF change _t | -0.015 | -0.013 | 0.008 | 0.008 | -0.24 | -3.16 | - 3.55 | - 3.43 | -4.06 | -0.06 |
| CFO_t | 0.049 | 0.070 | 0.079 | 0.064 | -4.16 | -5.65 | -3.18 | -2.08 | 1.49 | 3.49 |
| $CFO\ change_t$ | -0.017 | 0.005 | 0.028 | 0.001 | -4.89 | - 9.08 | -4.08 | -5.20 | 1.18 | 6.42 |
| CFI_t | 0.071 | 0.060 | 0.056 | 0.064 | 2.50 | 3.55 | 1.69 | 1.16 | -0.97 | -2.15 |
| CFI change _t | 0.026 | 0.000 | -0.020 | -0.002 | 5.07 | 8.25 | 5.73 | 3.76 | 0.46 | -3.52 |
| Funds from operations change _t | 0.004 | 0.008 | 0.012 | -0.002 | -1.31 | - 2.64 | 2.00 | - 1.64 | 3.82 | 4.88 |
| Working capital change _t | 0.021 | 0.003 | -0.016 | -0.003 | 4.22 | 7.86 | 5.84 | 4.45 | 1.68 | -3.14 |
| Observations | 2333 | 1557 | 1832 | 1755 | | | | | | |
| Germany | | | | | | | | | | |
| BL_t | 0.150 | 0.238 | 0.220 | 0.277 | -8.74 | -7.05 | -13.67 | 1.88 | -4.15 | -6.32 |
| $BLDev_t$ | -0.012 | -0.008 | 0.024 | 0.009 | -0.67 | - 7.41 | -4.53 | - 6.73 | - 3.81 | 3.32 |
| CH_{i} | 0.248 | 0.104 | 0.120 | 0.073 | 15.09 | 13.73 | 21.51 | -2.31 | 5.54 | 8.08 |
| CHD ev _t | 0.046 | 0.009 | -0.039 | -0.005 | 9.52 | 19.72 | 14.11 | 14.68 | 5.80 | -11.35 |
| $CH \ change_{t}$ | - 0.048 | -0.004 | 0.028 | 0.010 | -9.30 | - 14.63 | - 12.96 | -8.08 | -4.32 | 5.59 |
| CFF_{t} | -0.014 | -0.003 | -0.006 | 0.003 | -1.78 | -1.20 | -2.95 | 0.55 | -1.20 | -1.71 |
| CFF change _t | -0.026 | 0.004 | -0.003 | 0.008 | -3.14 | - 2.22 | - 3.83 | 0.79 | -0.71 | -1.42 |
| CFO_t | 0.045 | 0.061 | 0.075 | 0.069 | -2.22 | -4.29 | -3.69 | -2.46 | -1.54 | 1.22 |
| CFO change _t | - 0.003 | 0.000 | 0.035 | 0.006 | -0.43 | -5.17 | -1.42 | -5.63 | -1.17 | 5.20 |

| Table 6 (continued) | | | | | | | | | | |
|---|--------|--------|--------|--------|---------|---------|---------|---------|---------|---------|
| | A-L | A-S | B-L | B-S | t-stats | | | | | |
| | (1) | (2) | (3) | (4) | (1)–(2) | (1)–(3) | (1)-(4) | (2)–(3) | (2)–(4) | (3)–(4) |
| CFI_{l} | 0.075 | 0.063 | 0.045 | 0.061 | 2.02 | 5.06 | 2.63 | 3.65 | 0.57 | -3.35 |
| CFI change _t | 0.039 | -0.001 | -0.030 | -0.007 | 5.39 | 8.38 | 6.64 | 4.12 | 1.05 | -3.55 |
| Funds from operations change _t | 0.014 | 0.011 | 0.024 | 0.003 | 0.44 | - 0.85 | 2.49 | - 1.41 | 2.28 | 3.69 |
| Working capital change _t | 0.017 | 0.011 | -0.011 | -0.003 | 0.97 | 4.15 | 3.36 | 3.56 | 2.61 | -1.38 |
| Observations | 1574 | 1044 | 1315 | 1297 | | | | | | |
| Japan | | | | | | | | | | |
| BL_t | 0.195 | 0.257 | 0.240 | 0.263 | -15.65 | -11.08 | - 17.44 | 4.31 | -1.77 | -6.05 |
| $BLDev_t$ | -0.012 | -0.005 | -0.003 | -0.009 | -5.57 | - 6.06 | -2.17 | - 1.27 | 3.74 | 4.53 |
| CH_t | 0.220 | 0.139 | 0.146 | 0.113 | 32.39 | 28.88 | 44.67 | -3.40 | 14.30 | 17.36 |
| CHD ev _t | 0.028 | 0.005 | -0.032 | -0.007 | 24.63 | 54.73 | 39.18 | 44.33 | 21.04 | -29.43 |
| $CH \ change_t$ | -0.025 | -0.001 | 0.023 | 0.001 | -30.20 | -47.25 | -33.14 | -27.18 | -3.47 | 25.08 |
| CFF_t | -0.019 | -0.018 | -0.007 | -0.015 | -1.41 | -7.50 | -3.10 | -7.56 | -2.17 | 5.53 |
| CFF change _t | -0.015 | -0.001 | 0.010 | 0.004 | -7.86 | - 12.68 | - 10.99 | - 7.11 | - 4.24 | 3.38 |
| CFO_t | 0.040 | 0.052 | 0.058 | 0.050 | - 9.63 | -13.13 | -7.91 | -5.26 | 2.03 | 7.01 |
| $CFO\ change_t$ | -0.021 | 0.001 | 0.020 | 0.005 | -17.20 | -28.48 | - 19.40 | -15.37 | -2.90 | 12.64 |
| CFI_t | 0.044 | 0.034 | 0.029 | 0.032 | 8.27 | 11.50 | 9.63 | 4.67 | 1.83 | -2.94 |
| $CFI \ change_t$ | 0.013 | 0.003 | -0.007 | 0.003 | 7.62 | 13.50 | 7.47 | 8.29 | -0.20 | -8.47 |
| Funds from operations change _t | -0.004 | 0.004 | 0.011 | 0.002 | -5.53 | - 6.80 | - 0.99 | - 2.06 | 5.29 | 6.76 |
| Working capital change _t | 0.017 | 0.003 | -0.009 | -0.003 | 10.64 | 19.15 | 15.42 | 9.65 | 5.30 | -4.60 |
| Observations | 9332 | 5936 | 6652 | 6115 | | | | | | |
| UK | | | | | | | | | | |
| BL_{t} | 0.128 | 0.207 | 0.190 | 0.214 | -16.61 | -13.24 | - 19.59 | 3.53 | -1.43 | -5.33 |
| $BLDev_t$ | -0.018 | 0.004 | 0.022 | 0.008 | -8.15 | - 14.39 | -10.06 | - 6.47 | - 1.34 | 5.69 |
| CH_{t} | 0.244 | 0.104 | 0.126 | 0.071 | 26.18 | 22.99 | 37.46 | -5.55 | 10.58 | 16.40 |
| | | | | | | | | | | |

 Table 6
 (continued)

| | A-L | A-S | B-L | B-S | t-stats | | | | | |
|---|--------|--------|--------|--------|---------|---------|---------|---------|---------|---------|
| | (1) | (2) | (3) | (4) | (1)–(2) | (1)–(3) | (1)-(4) | (2)–(3) | (2)–(4) | (3)–(4) |
| CHDev _t | 0.050 | 0.012 | -0.044 | -0.005 | 15.53 | 34.36 | 24.90 | 23.88 | 10.57 | -18.79 |
| $CH \ change_t$ | -0.063 | 0.000 | 0.038 | 0.010 | -16.81 | -27.77 | -22.74 | - 14.57 | -5.63 | 12.28 |
| CFF_{t} | 0.016 | -0.007 | 0.026 | 0.009 | 4.55 | -1.77 | 1.46 | -6.21 | -3.90 | 3.44 |
| $CFF\ change_t$ | -0.033 | -0.022 | 0.001 | 0.011 | -1.39 | - 4.06 | - 6.18 | -3.18 | - 6.09 | -1.69 |
| CFO_t | 0.011 | 0.065 | 0.050 | 0.070 | -9.62 | -6.59 | -11.96 | 3.33 | -1.50 | -5.02 |
| $CFO\ change_t$ | -0.009 | 0.011 | 0.034 | 0.007 | -4.82 | -9.84 | -4.22 | -5.90 | 1.59 | 8.08 |
| CFI_t | 0.093 | 0.059 | 0.041 | 0.068 | 9.13 | 12.97 | 6.85 | 4.57 | -2.90 | -7.49 |
| $CFI \ change_t$ | 0.054 | -0.012 | -0.044 | 0.000 | 12.40 | 17.52 | 11.29 | 5.85 | -2.72 | - 9.04 |
| Funds from operations change _t | 0.00 | 0.007 | 0.022 | 0.007 | 0.51 | - 2.44 | 0.68 | -3.17 | 0.10 | 3.74 |
| Working capital change _t | 0.018 | -0.004 | -0.012 | 0.000 | 6.68 | 8.55 | 6.34 | 2.55 | 1.30 | -3.98 |
| Observations | 5536 | 3667 | 4541 | 4448 | | | | | | |
| SU | | | | | | | | | | |
| BL_{t} | 0.163 | 0.266 | 0.221 | 0.273 | -31.26 | -17.77 | -35.60 | 14.13 | -2.27 | -17.38 |
| $BLDev_t$ | -0.021 | -0.001 | 0.018 | 0.002 | - 11.47 | - 21.66 | - 13.74 | - 11.30 | - 2.08 | 9.72 |
| CH_{t} | 0.289 | 0.108 | 0.159 | 0.074 | 52.08 | 36.99 | 68.58 | -18.50 | 15.50 | 34.53 |
| CHDev _t | 0.047 | 0.010 | -0.041 | -0.005 | 26.41 | 54.36 | 39.76 | 39.53 | 18.26 | -29.89 |
| $CH \ change_t$ | -0.041 | 0.003 | 0.040 | 0.009 | -25.81 | -44.76 | -31.55 | -28.02 | -6.19 | 25.95 |
| CFF_{t} | 0.056 | 0.005 | 0.053 | 0.020 | 13.39 | 0.73 | 9.74 | -11.97 | -6.37 | 8.46 |
| CFF change _t | -0.039 | -0.007 | 0.019 | 0.016 | -6.32 | - 10.73 | - 11.61 | - 6.48 | - 7.99 | 0.95 |
| CFO_t | -0.015 | 0.073 | 0.048 | 0.073 | -19.01 | -12.32 | -20.49 | 6:59 | 0.10 | -7.08 |
| $CFO\ change_t$ | -0.014 | 0.014 | 0.040 | 0.005 | -8.51 | -14.34 | -6.54 | -8.45 | 4.64 | 12.57 |
| CFI_t | 0.086 | 0.074 | 0.060 | 0.084 | 4.28 | 9.15 | 0.77 | 6.27 | -4.92 | -10.78 |
| CFI change $_t$ | 0.026 | 0.006 | -0.010 | 0.010 | 5.41 | 9.12 | 4.42 | 4.98 | -2.00 | -7.07 |
| Funds from operations change _t | 0.002 | 0.014 | 0.029 | 0.005 | -3.21 | - 6.71 | -0.14 | - 4.87 | 5.53 | 8.93 |
| | | | | | | | | | | |

| continued) |
|------------|
| Ŭ |
| 9 |
| <u>م</u> |
| q |
| Ъ |

| | | | | | | 10,729 | 11,054 | 8994 | 13,851 | vations | Observ |
|---------|---------|---------|---------|---------|---------|--------|--------|-------|--------|--------------------------------|--------|
| -5.11 | 0.07 | 4.72 | 7.26 | 10.01 | 6.57 | 0.000 | -0.011 | 0.000 | 0.016 | ng capital change _t | Worki |
| (3)-(4) | (2)–(4) | (2)-(3) | (1)-(4) | (1)–(3) | (1)–(2) | (4) | (3) | (2) | (1) | | |
| | | | | | t-stats | B-S | B-L | A-S | A-L | | |
| | | | | | | | | | | | |

firms with above-target cash holdings and small deviations, B-L firms with below-target cash holdings and large deviations, and B-S firms with below-target cash holdings and small deviations in five countries—France, Germany, Japan, the US, Variables' definitions are provided in the "Appendix". *I*-statistics report the *t*-values of the This table summarizes the mean levels of major characteristics of different groups of firms: A-L firms with above-target cash holdings and large deviations from targets, A-S t-test for the hypothesis that the means for each pair of scenarios are equal not use it very productively while those with a large cash shortage experience a significant improvement in their levels of operational efficiency.

5.2 The asymmetry in the impact of firms' major characteristics on cash holdings adjustments

In another analysis, I examine the impact of several firm-specific factors such as precautionary motives of cash holdings, financial constraints, and corporate life-cycle on their cash holdings adjustments. These factors may affect their levels of costs of deviations, financial constraints, and costs of adjustments. For example, Sufi (2009) shows that firm size can be a strong statistical predictor of the use of bank lines of credit as large firms are likely to have more access to this source of financing. This suggests large firms may have weaker precautionary motives and be less responsive to deviations from target cash holdings due to lower costs of deviations from target cash holdings and hence experience slower speeds of adjustments. However, it can be also argued that as these firms are likely to face lower costs of adjustments due to better access to external financing sources, they can adjust toward their targets at lower costs, thus suggesting faster adjustments. Similarly, firms with low cash flow volatility may adjust either slowly for they are less likely to be affected by negative cash flow shocks (lower costs of deviations) or quickly as they tend to be matured firms with better access to external financing sources (lower costs of adjustments). To examine how these factors affect firms' target adjustments, I develop following partial, asymmetric cash holdings adjustment models:

$$\Delta CH_{it} = \lambda_0 + \lambda_1 CDev_{it}CH_{it}^L + \lambda_2 CDev_{it}CH_{it}^H + \varepsilon_{it}, \tag{5}$$

where $CH_{it}^{L}(CH_{it}^{H})$ is a dummy variable equal to 1 for firms with small size (CH_{it}^{SS}) , low cash flow volatility (CH_{it}^{LCFV}) , low dividend payout (CH_{it}^{LDPO}) , low growth opportunities (CH_{it}^{LGO}) or young age (CH_{it}^{Y}) (big size (CH_{it}^{BS}) , high cash flow volatility (CH_{it}^{HCFV}) , high dividend payout (CH_{it}^{HDPO}) , high growth opportunities (CH_{it}^{HGO}) or old age (CH_{it}^{O}) , and 0 otherwise.⁹ Next, I let these firm-specific characteristics interact with firms' deviations from target cash holdings to examine how they jointly determine firms' target adjustments. This may allow me to identify which factors have the first-order effects on these adjustments.

$$\Delta CH_{it} = \lambda_0 + \left(\lambda_1 CH_{it}^L + \lambda_2 CH_{it}^H\right) CDev_{it} CH_{it}^a + \left(\lambda_3 CH_{it}^L + \lambda_4 CH_{it}^H\right) CDev_{it} CH_{it}^b + \varepsilon_{it}.$$
(6)

The precautionary motives of cash holdings I first examine the impact of firm size by splitting the sample firms into two groups i.e., one with larger size and another with smaller size than the median size level. Panel A of Table 7 shows that large firms in Germany, Japan, the UK, and the US adjust at speeds which are statistically lower than those of small firms. This finding supports the argument that large firms with more access to external financing sources may be less concerned about deviations from targets but is contrary to the argument of Dittmar and Duchin (2010) that big firms with more access to

⁹ For example, in Panel A in Table 7, I divide the sample firms into two subgroups by using the median size level i.e., one with larger size (CH_{it}^{BS}) and another with smaller size (CH_{it}^{BS}) than the median size level of all firms. In Panel B, I also divide these firms into two subgroups but using the median level of cash flow volatility i.e., one with higher levels of cash flow volatility (CH_{it}^{HCFV}) and another with lower levels of cash flow volatility (CH_{it}^{LCFV}) than the median level of cash flow volatility.

| Table 7 Target cash holdings : | adjustments co | nditional on firn | ns' major charae | cteristics | | | | | | |
|--|----------------|-------------------|------------------|--------------|--------------|---------|---------|---------|---------|---------|
| | Coefficients | of speeds of ad | justments | | | t-stats | | | | |
| | (1) | (2) | (3) | (4) | (5) | (1) | (2) | (3) | (4) | (5) |
| | France | Germany | Japan | UK | SU | France | Germany | Japan | UK | SU |
| Panel A—firm size | | | | | | | | | | |
| $CDev_{ii} \cdot CH_{ii}^{SS}$ (i) | 0.416^{**} | 0.480^{**} | 0.462^{**} | 0.532^{**} | 0.542^{**} | (12.98) | (12.17) | (32.23) | (24.51) | (36.25) |
| $CDev_{ij} \cdot CH_{ij}^{BS}$ (ii) | 0.383^{**} | 0.320^{**} | 0.341^{**} | 0.470^{**} | 0.493 * * | (15.37) | (13.60) | (28.93) | (23.09) | (30.54) |
| F-test [(i) = (ii)] | 0.41 | 0.00 | 0.00 | 0.04 | 0.03 | | | | | |
| AR2 | 0.10 | 0.47 | 0.03 | 0.15 | 0.00 | | | | | |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^a_{ii}$ (i) | 0.515^{**} | 0.596^{**} | 0.529^{**} | 0.677^{**} | 0.590^{**} | (11.73) | (11.53) | (22.04) | (19.80) | (26.58) |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^B_{ii}$ (ii) | 0.490^{**} | 0.425** | 0.429^{**} | 0.591^{**} | 0.618^{**} | (11.95) | (11.97) | (23.21) | (18.53) | (26.68) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^{SS}_{ii}$ (iii) | 0.333^{**} | 0.372^{**} | 0.395** | 0.372^{**} | 0.538^{**} | (6.74) | (5.38) | (14.37) | (9.29) | (18.36) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^B_{ii}$ (iv) | 0.299^{**} | 0.212^{**} | 0.234^{**} | 0.325^{**} | 0.320^{**} | (7.37) | (4.49) | (11.94) | (8.81) | (11.52) |
| F-test [(i) = (ii)] | 0.67 | 0.00 | 0.00 | 0.04 | 0.37 | | | | | |
| F-test [(iii)=(iv)] | 0.56 | 0.02 | 0.00 | 0.34 | 0.00 | | | | | |
| F-test [(i)=(iii)] | 0.01 | 0.02 | 0.00 | 0.00 | 0.23 | | | | | |
| F-test [(ii)=(iv)] | 0.00 | 0.00 | 0.00 | 0.00 | 0.00 | | | | | |
| AR2 | 0.08 | 0.49 | 0.04 | 0.24 | 0.00 | | | | | |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 | | | | | |
| Panel B-cash flow volatility | | | | | | | | | | |
| $CDev_{ii} \cdot CH_{ii}^{LCFV}$ (i) | 0.360^{**} | 0.343^{**} | 0.376^{**} | 0.433^{**} | 0.471^{**} | (13.29) | (11.31) | (29.10) | (21.98) | (32.71) |
| $CDev_{ii} \cdot CH_{ii}^{HCFV}$ (ii) | 0.436^{**} | 0.463^{**} | 0.419^{**} | 0.495^{**} | 0.512^{**} | (15.20) | (14.08) | (34.03) | (25.78) | (38.00) |
| F-test [(i) = (ii)] | 0.04 | 0.01 | 0.01 | 0.02 | 0.04 | | | | | |
| AR2 | 0.10 | 0.39 | 0.02 | 0.14 | 0.00 | | | | | |
| $CDev_{ii} \cdot CH_{ii}^a \cdot CH_{ii}^{LCFV}$ (i) | 0.460^{**} | 0.482^{**} | 0.433^{**} | 0.551^{**} | 0.540^{**} | (11.59) | (10.83) | (21.61) | (16.94) | (28.32) |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^a_{ii} \cdot CH^{HCFV}_{ii}$ (ii) | 0.560^{**} | 0.548^{**} | 0.501^{**} | 0.605^{**} | 0.560^{**} | (13.28) | (11.69) | (24.72) | (20.38) | (29.02) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^b_{ii} \cdot CH^{LCFV}_{ii}$ (iii) | 0.287^{**} | 0.171^{**} | 0.326^{**} | 0.310^{**} | 0.378^{**} | (6.49) | (3.08) | (14.91) | (8.77) | (14.46) |
| $CDeV_{ii} \cdot CH_{ii}^b \cdot CH_{ii}^h CH_{ii}^{HCFV}$ (iv) | 0.333^{**} | 0.391^{**} | 0.322** | 0.354** | 0.455** | (7.83) | (6.83) | (13.49) | (9.93) | (18.45) |

| (continued) | |
|-------------|--|
| able 7 | |
| Table 7 | |

| | Coefficients | of speeds of adj | justments | | | t-stats | | | | |
|---|--------------|------------------|--------------|--------------|--------------|---------|---------|---------|---------|---------|
| | (1) | (2) | (3) | (4) | (5) | (1) | (2) | (3) | (4) | (5) |
| | France | Germany | Japan | UK | NS | France | Germany | Japan | UK | NS |
| F-test [(i)=(ii)] | 0.07 | 0.25 | 0.00 | 0.17 | 0.39 | | | | | |
| F-test [(iii)=(iv)] | 0.34 | 0.00 | 0.88 | 0.29 | 0.01 | | | | | |
| F-test [(i)=(iii)] | 0.01 | 0.00 | 0.00 | 0.00 | 0.00 | | | | | |
| F-test [(ii)=(iv)] | 0.00 | 0.06 | 0.00 | 0.00 | 0.00 | | | | | |
| AR2 | 0.09 | 0.45 | 0.03 | 0.25 | 0.00 | | | | | |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 | | | | | |
| Panel C-dividend payout | | | | | | | | | | |
| $CDev_{ii} \cdot CH_{ii}^{LDPO}$ (i) | 0.449** | 0.434^{**} | 0.425^{**} | 0.506^{**} | 0.523^{**} | (15.41) | (12.54) | (32.36) | (26.30) | (14.84) |
| $CDev_{ii} \cdot CH_{ii}^{HDPO}$ (ii) | 0.358^{**} | 0.396^{**} | 0.368^{**} | 0.411^{**} | 0.490^{**} | (13.10) | (12.98) | (29.35) | (20.89) | (51.86) |
| F-test $[(i) = (ii)]$ | 0.02 | 0.42 | 0.00 | 0.00 | 0.37 | | | | | |
| AR2 | 0.13 | 0.41 | 0.02 | 0.19 | 0.00 | | | | | |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^a_{ii} \cdot CH^{LDPO}_{ii}$ (i) | 0.527^{**} | 0.552^{**} | 0.525^{**} | 0.617^{**} | 0.639^{**} | (12.46) | (11.37) | (25.26) | (21.10) | (12.25) |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^a_{ii} \cdot CH^{HDPO}_{ii}$ (ii) | 0.467^{**} | 0.509^{**} | 0.412^{**} | 0.523^{**} | 0.540^{**} | (12.20) | (11.33) | (21.00) | (16.54) | (34.28) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^{LDPO}_{ii}$ (iii) | 0.367^{**} | 0.318^{**} | 0.308^{**} | 0.375^{**} | 0.404^{**} | (7.87) | (5.19) | (13.02) | (10.30) | (8.04) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^{HDPO}_{ii}$ (iv) | 0.284^{**} | 0.252^{**} | 0.332^{**} | 0.286^{**} | 0.435^{**} | (6.20) | (4.65) | (16.35) | (7.80) | (20.70) |
| F-test[(i)=(ii)] | 0.24 | 0.50 | 0.00 | 0.01 | 0.06 | | | | | |
| F-test [(iii)=(iv)] | 0.13 | 0.34 | 0.34 | 0.04 | 0.53 | | | | | |
| F-test [(i)=(iii)] | 0.02 | 0.01 | 0.00 | 0.00 | 0.00 | | | | | |
| F-test [(ii)=(iv)] | 0.01 | 0.00 | 0.01 | 0.00 | 0.00 | | | | | |
| AR2 | 0.11 | 0.42 | 0.03 | 0.30 | 0.00 | | | | | |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 | | | | | |
| Panel D-growth opportunitie | S | | | | | | | | | |
| $CDev_{ii}$ · CH_{ii}^{LGO} (i) | 0.387^{**} | 0.378^{**} | 0.456^{**} | 0.458** | 0.424** | (13.14) | (12.28) | (34.36) | (22.60) | (28.43) |
| | | | | | | | | | | |

| | Coefficients | s of speeds of ad | justments | | | t-stats | | | | |
|--|--------------|-------------------|--------------|--------------|--------------|---------|---------|---------|---------|---------|
| | (1) | (2) | (3) | (4) | (5) | (1) | (2) | (3) | (4) | (2) |
| | France | Germany | Japan | UK | SU | France | Germany | Japan | UK | SU |
| $CDev_{ii} \cdot CH_{ii}^{HGO}$ (ii) | 0.468** | 0.439^{**} | 0.373** | 0.482** | 0.543^{**} | (15.53) | (12.75) | (30.50) | (25.47) | (40.99) |
| F-test [(i)=(ii)] | 0.04 | 0.19 | 0.00 | 0.38 | 0.00 | | | | | |
| AR2 | 0.11 | 0.39 | 0.02 | 0.16 | 0.00 | | | | | |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^{LGO}_{ii}$ (i) | 0.487^{**} | 0.470** | 0.554^{**} | 0.556^{**} | 0.531^{**} | (10.22) | (10.17) | (26.82) | (17.22) | (22.89) |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^a_{ii} \cdot CH^{HGO}_{ii}$ (ii) | 0.555^{**} | 0.554^{**} | 0.436^{**} | 0.599^{**} | 0.547^{**} | (12.54) | (11.71) | (22.38) | (20.34) | (30.51) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^{LGO}_{ii}$ (iii) | 0.306^{**} | 0.281^{**} | 0.341^{**} | 0.354^{**} | 0.291^{**} | (6.31) | (5.47) | (15.29) | (10.26) | (11.74) |
| $CDev_{ii} \cdot CH_{ii}^b \cdot CH_{ii}^{b} \cdot CH_{ii}^{HGO}$ (iv) | 0.407^{**} | 0.332^{**} | 0.298^{**} | 0.334^{**} | 0.548^{**} | (8.26) | (5.45) | (13.25) | (8.96) | (21.04) |
| F-test [(i) = (ii)] | 0.28 | 0.16 | 0.00 | 0.28 | 0.55 | | | | | |
| F-test [(iii)=(iv)] | 0.10 | 0.43 | 0.12 | 0.63 | 0.00 | | | | | |
| F-test [(i)=(ii)] | 0.02 | 0.02 | 0.00 | 0.00 | 0.00 | | | | | |
| F-test [(ii)=(iv)] | 0.04 | 0.01 | 0.00 | 0.00 | 0.97 | | | | | |
| AR2 | 0.10 | 0.39 | 0.04 | 0.26 | 0.00 | | | | | |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 | | | | | |
| Panel E—firm age | | | | | | | | | | |
| $CDev_{ii} \cdot CH_{it}^{Y}(i)$ | 0.456** | 0.484^{**} | 0.501^{**} | 0.531^{**} | 0.528^{**} | (13.90) | (11.56) | (27.98) | (23.61) | (34.72) |
| $CDev_{ii} \cdot CH_{ii}^O$ (ii) | 0.358^{**} | 0.341^{**} | 0.326^{**} | 0.407 ** | 0.440^{**} | (13.39) | (14.03) | (30.11) | (23.14) | (32.85) |
| F-test [(i)=(ii)] | 0.02 | 0.00 | 0.00 | 0.00 | 0.00 | | | | | |
| AR2 | 0.10 | 0.44 | 0.03 | 0.18 | 0.00 | | | | | |
| $CDev_{ii} \cdot CH_{ii}^a \cdot CH_{ii}^a$ (i) | 0.586^{**} | 0.600^{**} | 0.593^{**} | 0.697^{**} | 0.603^{**} | (12.08) | (11.23) | (22.82) | (19.66) | (27.02) |
| $CDev_{ii} \cdot CH^a_{ii} \cdot CH^O_{ii}$ (ii) | 0.447^{**} | 0.450^{**} | 0.404^{**} | 0.471^{**} | 0.504^{**} | (11.32) | (12.28) | (23.56) | (15.04) | (24.84) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^Y_{ii}$ (iii) | 0.366^{**} | 0.394^{**} | 0.399^{**} | 0.319^{**} | 0.478^{**} | (6.51) | (5.80) | (13.25) | (7.13) | (17.03) |
| $CDev_{ii} \cdot CH^b_{ii} \cdot CH^O_{ii}$ (iv) | 0.272^{**} | 0.204^{**} | 0.237^{**} | 0.354^{**} | 0.373^{**} | (6.13) | (4.12) | (12.66) | (8.88) | (15.41) |
| F-test [(i)=(ii)] | 0.02 | 0.01 | 0.00 | 0.00 | 0.00 | | | | | |
| | | | | | | | | | | |

| (continued) | |
|-------------|--|
| Table 7 | |

| | Coefficients of | of speeds of adju | astments | | | t-stats | | | | |
|---------------------|-----------------|-------------------|----------|--------|--------|---------|---------|-------|-----|-----|
| | (1) | (2) | (3) | (4) | (5) | (1) | (2) | (3) | (4) | (5) |
| | France | Germany | Japan | UK | SU | France | Germany | Japan | UK | SU |
| F-test [(iii)=(iv)] | 0.17 | 0.01 | 0.00 | 0.55 | 0.00 | | | | | |
| F-test [(i)=(ii)] | 0.01 | 0.02 | 0.00 | 0.00 | 0.00 | | | | | |
| F-test [(ii)=(iv)] | 0.01 | 0.00 | 0.00 | 0.06 | 0.00 | | | | | |
| AR2 | 0.09 | 0.45 | 0.05 | 0.28 | 0.00 | | | | | |
| Observations | 7477 | 5230 | 28,035 | 18,192 | 44,628 | | | | | |
| | | | | | | | | | | |

the coefficient estimates for each pair of scenarios are equal. AR2 reports the p value of the test for no second-order serial correlation, which is asymptotically distributed as This table presents the SYS-GMM regression results for firms' asymmetric partial target adjustments conditional on levels of financial constraints, as modeled by Eqs. (5) and (6): $\Delta CH_{ii} = \lambda_0 + \lambda_1 CDev_{ii} CH_{ii}^L + \lambda_2 CDev_{ii} CH_{ii}^H + \varepsilon_{ii} \text{ and } \Delta CH_{ii} = \lambda_0 + (\lambda_1 CH_{ii}^L + \lambda_2 CH_{ii}^H) CDev_{ii} CH_{ii}^a + (\lambda_3 CH_{ii}^L + \lambda_4 CH_{ii}^H) CDev_{ii} CH_{ii}^b + \varepsilon_{ii}, \text{ where } \Delta CH_{ii} \text{ is the change}$ in cash holdings ratios. Target cash holdings are estimated by Eq. (2). See the "Appendix" for variables' definitions. CDevin is the difference between cash holdings in the last period and target cash holdings for the current period. CH_{ii}^{0} (CH_{ij}^{0}) is a dummy variable equal to 1 if cash holdings are higher than or equal to (lower than) targets and 0 otherwise. CH_{ii}^{H} (CH_{ii}^{R}) stands for $CH_{ii}^{BS}/CH_{ii}^{HCV}/CH_{ii}^{HOP}/CH_{ii}^{HCO}/CH_{ii}^{LCV}/CH_{ii}^{LOP}/CH_{ii}^{LO}/CH_{ii}^{V}$ which are dummy variables equal to 1 if firm size/cash flow volatility/ dividend payout/growth opportunities, and firm age is (are) greater than or equal to (lower than) the median level. My sample includes 103,562 firm-year observations for France, Germany, Japan, the UK and the US over the 1980–2007 period. Figures in parentheses are t-statistics. F-test reports the p value of the F-test for the hypothesis that V(0,1) under the null hypothesis of no serial correlation. ** and * indicate that the estimated coefficients are significant at the 1, and 5% levels of significance, respectively these sources may adjust faster. I have examined the mechanisms of adjustments of the two groups of firms but obtain results that do not form any clear patterns.

The interaction between firm size and whether firms have above- or below-target cash holdings does not change the above pattern as small firms (except for French firms) overall adjust toward their targets faster whether they experience above- or below-target cash holdings except for a few cases. For example, it turns out that in the presence of below-target (above-target) cash holdings, speeds of adjustments of small firms in the UK (the US) now do not statistically differ from those of large firms in the country.¹⁰

Next, I investigate the impact of firms' cash flow volatility on their adjustments by splitting the sample firms into two subgroups based on the median level of cash flow volatility. Firms with riskier cash flows may face higher costs of deviations from targets as they are more likely to experience negative cash flow shocks which may strongly affect their operations (e.g., their ability to meet debt obligations, finance working capital, and undertake growth opportunities) (Opler et al. 1999; Almeida et al. 2005; Bates et al. 2009). From Panel B of Table 7, there is strong evidence that these firms adjust faster than those with low cash flow volatility, which fits nicely with the argument that these firms are more likely to be affected by negative cash flow shocks, thus facing higher costs of deviations from targets. That evidence also supports Bakke and Gu (2017) who find that diversified firms tend to hold less cash and experience less cash flow volatility, thus possibly having lower target cash holdings and fewer incentives to adjust toward their targets.

Cash flow volatility has a less pronounced impact on firms' target cash holdings adjustments when this variable is interacted with firms' deviations from targets. Although by magnitude, firms with high cash flow volatility adjust faster whether they have above- or below-target cash holdings (except for firms with below-target cash holdings and high cash flow volatility in Japan), their speeds of adjustments do not statistically differ from those of firms with low cash flow volatility among firms with above-target cash holdings in France, Germany, the UK, and the US and those with below-target cash holdings in France, Japan, and the UK.

Dividend payout may also influence firms' adjustments as external investors may observe firms with higher dividend payout to be less risky, suggesting better access to capital markets and low costs of adjustments (Bates et al. 2009). Consistent with Bates et al. (2009) that the precautionary motive of cash holdings should be weaker for firms which pay more dividends, firms in France, Japan, and the UK which pay more dividends adjust statistically less quickly than those which pay less dividends (Panel C of Table 7).

When dividend payout is interacted with firms' deviations from targets, I find some different results. For example, speeds of adjustments for French firms with high dividend payout do not statistically differ from those of firms with low dividend payout among both groups of firms with above- and those with below-target cash holdings although by magnitude, firms with low dividend payout still adjust faster. This is a similar story for firms with below-target cash holdings in Japan.

Financial constraints Financial constraints i.e., growth opportunities and dividend payout may also influence firms' cash holdings adjustments. As I have discussed the impact of dividend payout, here I only focus on that of growth opportunities. Almeida et al. (2002) suggest that the benefit of achieving optimal cash holdings may be higher for high-growth firms since it helps them avoid situations where a cash shortage forces them to give up

¹⁰ Controlling for size and other firm-specific characteristics, I find strong evidence that firms with abovetarget cash holdings adjust statistically faster than those with below-target cash holdings.

positive *NPV* investments. Agency theories, however, suggest that these firms tend to be subject to more information asymmetries which lead to less access to external capital markets i.e., less ability to adjust via *CFF*.

Panel D of Table 7 reveals that by magnitude, except for firms in Japan, firms in all other countries experience faster speeds of adjustments when having more growth opportunities. High-growth firms in France and the US adjust statistically faster than their low-growth counterparts. The evidence for Japanese firms is somewhat puzzling as the presence of limited growth opportunities leads to faster adjustments. The interaction between growth opportunities. For example, the presence of more growth opportunities does not lead to statistically faster speeds of adjustments among both firms with above-target cash holdings in France and the US and those with below-target cash holdings in France.

Corporate life-cycle Finally, I investigate how firms' age may influence their target cash holdings adjustments since it can be reasonably considered as a proxy for their reputation and access to external financing sources (Diamond 1991). Dittmar and Duchin (2010) show that as firms' age increases, the precautionary motive of cash holdings may become weaker. It is therefore likely that matured firms may hold less cash and adjust less quickly than younger firms. To test the impact of firms' age, I divide the sample firms into two subsamples—young firms whose age is less than and matured firms whose age is higher than the median age level.

Panel E of Table 7 shows strong evidence across firms in five countries that the precautionary motive of cash holdings may be weaker for matured firms. In particular, matured firms adjust toward their targets at speeds from 0.326 (Japanese firms) to 0.440 (US firms), statistically slower than those for young firms (from 0.456 (French firms) to 0.531 (UK firms)). Letting firms' age interact with their deviations from targets, I find that young firms adjust faster whether they experience above- or below-target cash holdings, except for a few cases i.e., speeds of adjustments for young firms in France and the UK are not statistically different from those for matured firms in these two countries in the presence of below-target cash holdings. Young firms with above-target cash holdings experience fastest speeds of adjustments among the four groups of firms.

Overall, I show that firms' precautionary motives of cash holdings, financial constraints, and corporate life-cycle may generally influence their cash holdings adjustments although their impact tends to become less significant when I interact these factors with deviations from target cash holdings (except for firm age). The impact of deviations from target cash holdings, however, remains strong and significant even when these factors are taken into account. This suggests that firms' deviations from target cash holdings may be the major driver of their target adjustments, not firm-specific characteristics.

6 Concluding remarks

Consistent with the trade-off view of corporate cash holdings, I find that there is asymmetry in not only firms' speeds but also mechanisms of adjustments. In particular, firms with above-target cash holdings generally disgorge their excess cash by reducing *CFF* (across firms in all the sample countries) and *CFO* (except for firms in the UK and the US) but increasing *CFI* (across firms in all the sample countries) while those with below-target cash holdings undertake opposite mechanisms to build up their cash reserves. Since the mechanisms undertaken by the former firms may incur lower costs, they can adjust

relatively faster than the latter firms. Further, firms with above-target cash holdings adjust toward their targets mainly through changes in *CFF* and *CFI* while those with below-target cash holdings undertake major adjustments in *CFO*. The evidence on the asymmetry in the adjustments in *CFO* and *CFI* undertaken by these two groups of firms is particularly novel. Previous studies in the area tend to treat adjustments in firms' *CFO* and *CFI* as exogenous, independent of firms' cash and leverage management policies but I now explicitly account for these to get a better insight into firms' cash holdings behaviors.

In my additional analyses, I find evidence that contrary to firms with small deviations from target cash holdings, those with large deviations may undertake faster adjustments due to lower costs of adjustments and higher costs of deviations, a finding in favor of both the "shared" fixed cost argument and agency theories. Intriguingly, I find that, consistent with agency theories, firms with a lot of excess cash may not use it very effectively. This problem is particularly prominent among firms with poor corporate governance (French and Japanese firms in particular). However, when having less cash than optimal levels, firms try hard to improve their operational efficiency.

Finally, I document some evidence on the impact of firms' precautionary motives of cash holdings, financial constraints, and corporate life-cycle on their target cash holdings adjustments. I find evidence that firms adjust toward their target cash holdings faster contingent on having smaller size, higher cash flow volatility, lower dividend payout, more growth opportunities, and younger age. When these factors are interacted with firms' deviations from targets, however, their influence on firms' target adjustments tends to become less significant in most cases (except for firm age), suggesting that firms' deviations from targets may be a more important driver of their target cash holdings adjustments.

Overall, the study contributes to the cash holdings literature by focusing on the asymmetry in the mechanisms undertaken by firms to move toward their target cash holdings levels. The findings complement the existing evidence on the asymmetry in firms' speeds of adjustments reported by previous studies within this literature.

Open Access This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

| Variable | Definition |
|----------------------------------|---|
| Asset tangibility (AT) | Fixed assets scaled by the book value of total assets |
| Balance sheet net debt (BSND) | The difference between the book value of total debt and cash and cash equivalents scaled by the book value of total assets |
| Book leverage (BL) | The ratio of the book value of total debt to the book value of total assets |
| Cash (CH) | Cash and cash equivalents scaled by the book value of total assets |
| Cash dividends | Cash dividends paid scaled by the book value of total assets |
| Cash holdings change (CH change) | The difference between cash holdings in the current and the last accounting periods scaled by the book value of total assets |

Appendix: Variable definition

The asymmetry in firms' mechanisms of cash holdings adjustments:...

| Variable | Definition |
|-----------------------------------|---|
| Cash flows (CF) | Operating income before depreciation minus interest expense, income taxes and total dividends paid scaled by the book value of total assets |
| Cash flows from financing (CFF) | Net cash flow from financing activities (net cash receipts and disburse- ments resulting from reduction and/or increase in long or short term debt, proceeds from sale of stock, stock repurchased/redeemed/ retired, dividends paid and other financing activities) scaled by the book value of total assets |
| Cash flows from investing (CFI) | Net cash flow from investing activities (net cash receipts and disburse- ments resulting from capital expenditures, decrease/increase from investments, disposal of fixed assets, increase in other assets and other investing activities) scaled by the book value of total assets |
| Cash flows from operating (CFO) | Net cash flow from operating activities (net cash receipts and disburse- ments resulting from the operations of the firm i.e., the sum of funds from operations, funds from/used for other operating activities and extraordinary items) scaled by the book value of total assets |
| Cash flow volatility (CFV) | The absolute value of the difference between the first difference of cash flow (% change) and the average of first differences |
| Capital expenditures (CAPEX) | Capital expenditures scaled by the book value of total assets |
| Equity proceeds | Net proceeds from sale/issue of common and preferred equity scaled by the book value of total assets |
| Equity repurchases | Common/preferred equity purchased, retired, converted, and redeemed scaled by the book value of total assets |
| Firm size (FS) | The natural log of the book value of total assets measured in 1980 US\$ value |
| Growth opportunities (GO) | The market-to-book ratio (the market value of total assets (the sum of the market value of equity or market capitalization and the book value of total debt) scaled by the book value of total assets) |
| Market leverage (ML) | The ratio of the book value of total debt to the market value of total assets (the sum of the market value of equity (market capitalization) plus the book value of total debt) |
| Net debt issue | The sum of net long-term debt issue (long-term borrowings minus long-term debt reductions) and increase/decrease in short-term bor- rowing scaled by the book value of total assets |
| Net equity issue | The net difference between equity proceeds and equity repurchases |
| Net working capital (NWC) | The difference between working capital and cash and cash equivalents scaled by the book value of total assets |
| Research and development $(R\&D)$ | Research and development expenses scaled by total sales |
| Working capital | Funds from/for other operating activities scaled by the book value of total assets |

References

Allen F, Michaely R (2003) Chapter 7—payout policy. In: Constantinides GM, Harris M, Stulz RM (eds) Handbook of the economics of finance. Elsevier, Amsterdam, pp 337–429

Allen F, Bernardo AE, Welch I (2002) A theory of dividends based on tax clienteles. J Finance 55:2499–2536

Almeida H, Campello M, Weisbach MS (2002) Corporate demand for liquidity. Working Paper, University of Illinois and New York University

Almeida H, Campello M, Weisbach MS (2005) The cash flow sensitivity of cash. J Finance 59:1777-1804

- Ando A, Christelis D, Miyagawa T (2003) Inefficiency of corporate investment and distortion of savings behavior in Japan. Struct. In: Blomström M, Corbett J, Hayashi F, Kashyap A (eds) Structural Impediments to Growth in Japan, Conference held Tokyo, Japan, University of Chicago Press, pp 155–190
- Baker M, Wurgler J (2010) Signaling with reference points: behavioral foundations for the Lintner model of dividends. Working Paper, New York University
- Baker M, Nagel S, Wurgler J (2007) The effect of dividends on consumption. Brook Pap Econ Act Econ Stud Progr 38(1):231–292
- Bakke T-E, Gu T (2017) Diversification and cash dynamics. J Finance Econ 123:580-601
- Bates TW, Kahle KM, Stulz RM (2009) Why do U.S. firms hold so much more cash than they used to? J Finance 64:1985–2021
- Baumol WJ (1952) The transactions demand for cash: an inventory theoretic approach. Q J Econ 66:545-556
- Best RJ, Best RW (2001) Prior information and the market reaction to dividend changes. Rev Quant Finance Acc 17:361–376
- Bhattacharya S (1979) Imperfect information, dividend policy, and the 'bird in the hand' fallacy. Bell J Econ 10:259–270
- Bliss RT, Rosen RJ (2001) CEO compensation and bank mergers. J Finance Econ 61:107-138
- Brav A, Graham JR, Harvey CR, Michaely R (2005) Payout policy in the 21st century. J Finance Econ 77:483–527
- Brealey RA, Myers SC, Allen F (2014) Principles of corporate finance, 11th edn. McGraw-Hill/Irwin, New York
- Brennan MJ, Thakor AV (1990) Shareholder preferences and dividend policy. J Finance 45:993–1018

Brown JR, Petersen BC (2011) Cash holdings and R&D smoothing. J Corp Finance 17:694-709

- Byoun S (2008) How and when do firms adjust their capital structures toward targets? J Finance 63:3069–3096
- Chiang W, Monahan GE (2005) Managing inventories in a two-echelon dual-channel supply chain. Eur J Oper Res 162:325–341
- DeAngelo H, DeAngelo L (2007) Capital structure, payout policy, and financial flexibility. Working Paper, University of Southern California
- DeMarzo PM, Sannikov Y (2008) Learning in dynamic incentive contracts. Working Paper, Stanford University
- Diamond DW (1991) Monitoring and reputation: the choice between bank loans and directly placed debt. J Pol Econ 99:689-721
- Dittmar A, Duchin R (2010) The dynamics of cash. Working Paper, University of Michigan
- Dittmar A, Mahrt-Smith J (2007) Corporate governance and the value of cash holdings. J Finance Econ 83:599–634
- Faulkender M, Flannery MJ, Hankins KW, Smith JM (2012) Cash flows and leverage adjustments. J Finance Econ 103:632–646
- Fazzari SM, Hubbard GR, Petersen BC (1988) Financing constraints and corporate investment. Brook Pap Econ Act 1:141–195
- Fudenberg D, Tirole J (1995) A theory of income and dividend smoothing based on incumbency rents. J Pol Econ 103:75–93
- García-Teruel PJ, Martínez-Solano P (2010) A dynamic perspective on the determinants of accounts payable. Rev Quant Finance Acc 34:439–457
- Guttman I, Kadan O, Kandel E (2010) Dividend stickiness and strategic pooling. Rev Finance Stud 23:4455-4495
- Harford J (2002) Corporate cash reserves and acquisitions. J Finance 54:1969–1997
- Harford J, Li KAI (2007) Decoupling CEO wealth and firm performance: the case of acquiring CEOs. J Finance 62:917–949
- Harford J, Mansi SA, Maxwell WF (2012) Corporate governance and firm cash holdings. In: Boubaker S, Nguyen BD, Nguyen DK (eds) Corporate governance—recent developments and new trends. Springer, Berlin, pp 107–138
- Iyer SR, Feng H, Rao RP (2017) Payout flexibility and capital expenditure. Rev Quant Finance Acc 49:633–659
- Jensen MC (1986) Agency cost of free cash flows, corporate finance, and takeovers. Am Econ Rev 76:323-329
- Jiang Z, Lie E (2016) Cash holding adjustments and managerial entrenchment. J Corp Finance 36:190-205
- John K, Williams J (1985) Dividends, dilution, and taxes: a signaling equilibrium. J Finance 40:1053–1070
- Kumar P (1988) Shareholder-manager conflict and the information content of dividends. Rev Finance Stud 1:111–136

Kumar P, Lee B-S (2001) Discrete dividend policy with permanent earnings. Finance Manag 30:55–76

Lambrecht BM, Myers SC (2010) A Lintner model of dividends and managerial rents. Working Paper, MIT Sloan School

- Leary MT, Roberts MR (2005) Do firms rebalance their capital structure? J Finance 60:2575-2619
- Lee BS, Suh J (2011) Cash holdings and share repurchases: international evidence. J Corp Finance 17:1306–1329
- Leftwich R, Zmijewski ME (1994) Contemporaneous announcements of dividends and earnings. J Acc Audit Finance 9:725–762
- Miller MH, Modigliani F (1961) Dividend policy, growth and the valuation of shares. J Bus 34:411-433
- Miller MH, Orr D (1966) A model of the demand for money by firms. Q J Econ 80:413-435
- Miller MH, Rock K (1985) Dividend policy under asymmetric information. J Finance 40:1031-1051
- Miller MH, Scholes MS (1978) Dividends and taxes. J Financ Econ 6:333-364
- Minton BA, Schrand C (1999) The impact of cash flow volatility on discretionary investment and the costs of debt and equity financing. J Finance Econ 54:423–460
- Mulligan CB (1997) Scale economies, the value of time, and the demand for money: longitudinal evidence from firms. J Pol Econ 105:1061–1079
- Myers SC, Majluf NS (1984) Corporate financing and investment decisions when firms have information that investors do not have. J Finance Econ 13:187–221
- Opler T, Pinkowitz L, Stulz R, Williamson R (1999) The determinants and implications of corporate cash holdings. J Finance Econ 52:3–46
- Orlova SV, Rao RP (2018) Cash holdings speed of adjustment. Int Rev Econ Finance 54:1-14
- Ozkan A, Ozkan N (2004) Corporate cash holdings: an empirical investigation of UK companies. J Bank Finance 28:2103–2134
- Öztekin Ö (2015) Capital structure decisions around the world: which factors are reliably important? J Finance Quant Anal 50:301–323
- Riddick LA, Whited TM (2009) The corporate propensity to save. J Finance 64:1729–1766
- Strebulaev IA (2007) Do tests of capital structure theory mean what they say? J Finance 62:1747–1787
- Sufi A (2009) Bank lines of credit in corporate finance: an empirical analysis. Rev Finance Stud 22:1057–1088
- Teng M, Hachiya T (2013) Agency problems and stock repurchases: evidence from Japan. Rev Pac Basin Finance Mark Pol 16:1–30
- Venkiteshwaran V (2011) Partial adjustment toward optimal cash holding levels. Rev Finance Econ 20:113-121
- Wu W, Rui OM, Wu C (2012) Trade credit, cash holdings, and financial deepening: Evidence from a transitional economy. J Bank Finance 36:2868–2883